Neuroprotective Effects of Phytochemicals in Neurological Disorders



EDITORS Tahira Farooqui and Akhlaq A. Farooqui

WILEY Blackwell

Neuroprotective Effects of Phytochemicals in Neurological Disorders

Neuroprotective Effects of Phytochemicals in Neurological Disorders

Edited by Tahira Farooqui and Akhlaq A. Farooqui

WILEY Blackwell

Copyright © 2017 by John Wiley & Sons, Inc. All rights reserved

Published by John Wiley & Sons, Inc., Hoboken, New Jersey

Published simultaneously in Canada

No part of this publication may be reproduced, stored in a retrieval system, or transmitted in any form or by any means, electronic, mechanical, photocopying, recording, scanning, or otherwise, except as permitted under Section 107 or 108 of the 1976 United States Copyright Act, without either the prior written permission of the Publisher, or authorization through payment of the appropriate per-copy fee to the Copyright Clearance Center, Inc., 222 Rosewood Drive, Danvers, MA 01923, (978) 750-8400, fax (978) 750-4470, or on the web at www.copyright.com. Requests to the Publisher for permission should be addressed to the Permissions Department, John Wiley & Sons, Inc., 111 River Street, Hoboken, NJ 07030, (201) 748-6011, fax (201) 748-6008, or online at http://www.wiley.com/go/permissions.

Limit of Liability/Disclaimer of Warranty: While the publisher and author have used their best efforts in preparing this book, they make no representations or warranties with respect to the accuracy or completeness of the contents of this book and specifically disclaim any implied warranties of merchantability or fitness for a particular purpose. No warranty may be created or extended by sales representatives or written sales materials. The advice and strategies contained herein may not be suitable for your situation. You should consult with a professional where appropriate. Neither the publisher nor author shall be liable for any loss of profit or any other commercial damages, including but not limited to special, incidental, consequential, or other damages.

For general information on our other products and services or for technical support, please contact our Customer Care Department within the United States at (800) 762-2974, outside the United States at (317) 572-3993 or fax (317) 572-4002.

Wiley also publishes its books in a variety of electronic formats. Some content that appears in print may not be available in electronic formats. For more information about Wiley products, visit our web site at www.wiley.com.

Library of Congress Cataloging-in-Publication Data

Names: Farooqui, Tahira, editor. | Farooqui, Akhlaq A., editor.
Title: Neuroprotective effects of phytochemicals in neurological disorders / edited by Tahira Farooqui, Akhlaq Farooqui.
Description: Hoboken, New Jersey : John Wiley & Sons, Inc., [2017] | Includes bibliographical references and index.
Identifiers: LCCN 2016033001 (print) | LCCN 2016034179 (ebook) | ISBN 9781119155140 (cloth) | ISBN 9781119155171 (pdf) | ISBN 9781119155188 (epub)
Subjects: | MESH: Nervous System Diseases-drug therapy | Phytochemicals-pharmacology | Phytochemicals-therapeutic use | Neuroprotective Agents-pharmacology | Neuroprotective Agents-therapeutic use
Classification: LCC RM315 (print) | LCC RM315 (ebook) | NLM WL 140 | DDC 616.8/0461-dc23 LC record available at https://lccn.loc.gov/2016033001
Cover Images: spices @ dianaduda / Shutterstock;

citrus fruits © Luke Michael / Unsplash.com; brainstem © MedicalRD.com / Getty Images; market © sydeen / Shutterstock.com

Set in 10/12pt Warnock by SPi Global, Pondicherry, India

10 9 8 7 6 5 4 3 2 1

Phytochemicals hold a special, elite place in the nutritional landscape. Joel Fuhrman, MD

Contents

Contributors *xi* Preface *xix* Acknowledgments *xxiii*

- 1 Use of Phytochemicals against Neuroinflammation 1 Wei-Yi Ong, Tahira Farooqui, Christabel Fung-Yih Ho, Yee-Kong Ng, and Akhlaq A. Farooqui
- 2 Flavonoids in Transgenic Alzheimer's Disease Mouse Models: Current Insights and Future Perspectives 43 Angélica Maria Sabogal-Guáqueta, Edison Osorio, and Gloria Patricia Cardona-Gómez
- 3 Neuroprotective Effects of Polyphenols in Aging and Age-Related Neurological Disorders 65 Giulia Corona and David Vauzour
- 4 Indian Herbs and their Therapeutic Potential against Alzheimer's Disease and Other Neurological Disorders 79 Navrattan Kaur, Bibekananda Sarkar, Iqbal Gill, Sukhchain Kaur, Sunil Mittal, Monisha Dhiman, Prasad R. Padala, Regino Perez-Polo, and Anil K. Mantha
- 5 Garlic and its Effects in Neurological Disorders 113 Akhlaq A. Farooqui and Tahira Farooqui
- **6** Effects of Extra-Virgin Olive Oil in Neurological Disorders 133 Akhlaq A. Farooqui and Tahira Farooqui
- 7 Ginger Components as Anti-Alzheimer Drugs: Focus on Drug Design 149 Faizul Azam
- 8 Phytomedicine: A Possible Tool for Alzheimer's Disease Therapeutics 167 Jai Malik
- 9 Effects of Phytochemicals on Diabetic Retino-neuropathy 199 Mohammad Shamsul Ola, Mohd Imtiaz Nawaz, and Abdullah S. Alhomida

- viii Contents
 - 10 Herbal Drugs: A New Hope for Huntington's Disease 213 Jai Malik
 - 11 Neuroprotective Properties of Dietary Polyphenols in Parkinson's Disease 243 Altaf S. Darvesh, McKenna McClure, Prabodh Sadana, Chris Paxos, Werner J. Geldenhuys, Joshua D. Lambert, Tariq M. Haqqi, and Jason R. Richardson
 - 12 Potential of Polyphenols in the Treatment of Major Depression: Focus on Molecular Aspects 265 Ashish Dhir
 - 13 Effect of Phytochemicals on Diabetes-Related Neurological Disorders 283 Abubakar Mohammed, Kanti Bhooshan Pandey, and Syed Ibrahim Rizvi
 - 14 Neuroprotective Effects of Extra-Virgin Olive Oil and its Components in Alzheimer's Disease 299 Alaa H. Abuznait, Hisham Qosa, Loqman A. Mohamed, Yazan S. Batarseh, and Amal Kaddoumi
 - 15 Protective Role of Black-Tea Extract in a Transgenic Drosophila Model of Parkinson's Disease 317 Yasir Hasan Siddiaue
 - 16 Apitherapy: Therapeutic Effects of Propolis on Neurological Disorders 335 Tahira Farooqui and Akhlag A. Farooqui
 - 17 Molecular Mechanisms behind the Beneficial Activity of Polyunsaturated Fatty Acids in Alzheimer's Disease and Related Conditions 359 Undurti N. Das
 - 18 Prevention of Neuroinflammation by Resveratrol: Focus on Experimental Models and Molecular Mechanisms 377 Justine Renaud and Maria-Grazia Martinoli
 - 19 Modulation of the Estradiol and Neprilysin Pathways by Resveratrol in a Lipopolysaccharide Model of Cognitive Impairment 395 Nesrine S. El Sayed
 - 20 Neuroprotective Effect of Resveratrol in Cerebral Ischemia 407 Nilanjan Ghosh, Rituparna Ghosh, Subhash C. Mandal, and Mahadeb Pal
 - 21 Effects of Nobiletin in Animal Models of Cognitive Impairment: Current Insights and Future Perspectives 421 Akira Nakajima, Yasushi Ohizumi, and Kiyofumi Yamada
 - 22 Potential Neuroprotective Effects of Curcumin against Dementia 435 Natascia Brondino, Laura Fusar-Poli, Cristina Panisi, and Pierluigi Politi

- 23 Neuroprotective Activity of Curcumin and *Emblica officinalis* Extract against Carbofuran-Induced Neurotoxicity in Wistar Rats 447 *Ramadasan Kuttan and P.P. Binitha*
- **24 Potential Use of Curcuminoids for the Treatment of Alzheimer's Disease** 463 Tougeer Ahmed, Sana Javed, Ameema Tariq, and Anwarul-Hassan Gilani
- 25 Prevention by Curcumin of Neuroinflammation in Intracerebral Hemorrhage 489 Yujie Chen and Hua Feng
- **26 Effect of Polyphenols on Protein Misfolding** 501 Rona Banerjee
- 27 Molecular Mechanisms Involved in the Neuroprotective Action of Phytochemicals 515 Aditya Sunkaria, Aarti Yadav, Sunil Kumar Sharma, and Rajat Sandhir
- **28** Nutraceuticals and Cognitive Dysfunction: Focus on Alzheimer's Disease 561 Virginia Boccardi, Clara Tinarelli, and Patrizia Mecocci
- **29 Summary and Perspective** 581 Tahira Farooqui and Akhlaq A. Farooqui

Index 595

Contributors

Alaa H. Abuznait

Department of Basic Pharmaceutical Sciences School of Pharmacy University of Louisiana at Monroe Monroe, LA USA

Tougeer Ahmed

Atta-ur-Rahman School of Applied Biosciences National University of Sciences and Technology Islamabad Pakistan

Abdullah S. Alhomida

Department of Biochemistry College of Science King Saud University Riyadh KSA

Faizul Azam

Faculty of Pharmacy Misurata University Misurata Libya

Rona Banerjee

Indian Institute of Technology Roorkee India

Yazan S. Batarseh

Department of Basic Pharmaceutical Sciences School of Pharmacy University of Louisiana at Monroe Monroe, LA USA

P.P. Binitha

Amala Cancer research Centre Amala Nagar Thrissur Kerala India

Virginia Boccardi

Institute of Gerontology and Geriatrics Department of Medicine University of Perugia Perugia Italy

Natascia Brondino

Department of Brain and Behavioral Sciences University of Pavia Pavia Italy

Gloria Patricia Cardona-Gómez

Cellular and Molecular Neurobiology Area Neuroscience Group of Antioquia School of Medicine SIU University of Antioquia Medellín Colombia xii Contributors

Yujie Chen

Department of Neurosurgery Southwest Hospital Third Military Medical University Chongqing China

Giulia Corona

Health Sciences Research Centre University of Roehampton London UK

Undurti N. Das

UND Life Sciences Federal Way, WA USA

Altaf S. Darvesh Department of Pharmaceutical Sciences College of Pharmacy Neurodegenerative Diseases and Aging Focus Group and Department of Psychiatry College of Medicine Northeast Ohio Medical University Rootstown, OH USA

Monisha Dhiman

Centre for Biochemistry and Microbial Sciences School of Basic and Applied Sciences Central University of Punjab Bathinda India

Ashish Dhir

Department of Neurology School of Medicine University of California Davis Sacramento, CA USA

Nesrine S. El Sayed

Department of Pharmacology and Toxicology Faculty of Pharmacy Cairo University and Department of Pharmacology and Toxicology Faculty of Pharmacy and Biotechnology German University in Cairo Cairo Egypt

Akhlaq A. Farooqui

Department of Molecular and Cellular Biochemistry Ohio State University Columbus, OH USA

Tahira Farooqui

Department of Entomology Ohio State University Columbus, OH USA

Hua Feng

Department of Neurosurgery Southwest Hospital Third Military Medical University Chongqing China

Laura Fusar-Poli

Department of Brain and Behavioral Sciences University of Pavia Pavia Italy

Werner J. Geldenhuys

Department of Pharmaceutical Sciences School of Pharmacy Robert C. Byrd Health Sciences Center West Virginia University Morgantown, WV USA

Anwarul-Hassan Gilani

Natural Products Research Unit Department of Biological and Biomedical Sciences The Aga Khan University Medical College Karachi and Pakistan Council for Science and Technology Islamabad Pakistan

Iqbal Gill

Centre for Animal Sciences School of Basic and Applied Sciences Central University of Punjab Bathinda India

Tariq M. Haqqi

Department of Anatomy & Neurobiology College of Medicine Northeast Ohio Medical University Rootstown, OH USA

Christabel Fung-Yih Ho

Department of Anatomy National University of Singapore Singapore

Sana Javed

Atta-ur-Rahman School of Applied Biosciences National University of Sciences and Technology Islamabad Pakistan

Amal Kaddoumi

Department of Basic Pharmaceutical Sciences School of Pharmacy University of Louisiana at Monroe Monroe, LA USA

Navrattan Kaur

Centre for Animal Sciences School of Basic and Applied Sciences Central University of Punjab Bathinda India

Sukhchain Kaur

Centre for Biochemistry and Microbial Sciences School of Basic and Applied Sciences Central University of Punjab Bathinda India

Ramadasan Kuttan

Amala Cancer research Centre Amala Nagar Thrissur Kerala India

Joshua D. Lambert

Department of Food Science College of Agricultural Sciences The Pennsylvania State University University Park, PA USA

Jai Malik

University Institute of Pharmaceutical Sciences UGC Centre of Advanced Study Panjab University Chandigarh India

Anil K. Mantha

Centre for Animal Sciences School of Basic and Applied Sciences Central University of Punjab Bathinda India xiv Contributors

Maria-Grazia Martinoli

Department of Medical Biology and Cellular Signalling Research Group University of Quebec Trois-Rivières, QC and Department of Psychiatry and Neuroscience Laval University and CHU Research Center Quebec City, QC Canada

McKenna McClure

Department of Pharmaceutical Sciences College of Pharmacy Neurodegenerative Diseases and Aging Focus Group Northeast Ohio Medical University Rootstown, OH USA

Patrizia Mecocci

Institute of Gerontology and Geriatrics Department of Medicine University of Perugia Perugia Italy

Sunil Mittal

Centre for Environmental Science and Technology School of Environment and Earth Sciences Central University of Punjab Bathinda India

Loqman A. Mohamed

Department of Basic Pharmaceutical Sciences School of Pharmacy University of Louisiana at Monroe Monroe, LA USA

Abubakar Mohammed

Department of Biochemistry University of Allahabad Allahabad India

Akira Nakajima

Department of Neuropsychopharmacology and Hospital Pharmacy Graduate School of Medicine Nagoya University Showa-ku, Nagoya Japan

Mohd Imtiaz Nawaz

Department of Molecular and Translational Medicine University of Brescia Brescia Italy

Yee-Kong Ng

Department of Anatomy National University of Singapore Singapore

Nilanjan Ghosh

Dr. B.C. Roy College of Pharmacy and Allied Health Sciences Durgapur India

Yasushi Ohizumi

Department of Medical Biochemistry School of Pharmaceutical Sciences University of Shizuoka Suruga-ku, Shizuoka and Kansei Fukushi Research Institute Tohoku Fukushi University Sendai Japan

Mohammad Shamsul Ola

Department of Biochemistry College of Science King Saud University Riyadh KSA

Wei-Yi Ong

Department of Anatomy and Neurobiology and Ageing Research Programme National University of Singapore Singapore

Edison Osorio

Bioactive Substances Research Group Faculty of Pharmaceutical and Food Science University of Antioquia Medellín Colombia

Prasad R. Padala

Geriatric Research, Education and Clinical Center Central Arkansas Veterans Healthcare System Little Rock, AR USA and Donald W. Reynolds Department of Geriatrics and Department of Psychiatry University of Arkansas for Medical Sciences Little Rock, AR USA

Mahadeb Pal

Division of Molecular Medicine Bose Institute Kolkata India

Kanti Bhooshan Pandey

Marine Biotechnology and Ecology Division CSIR – Central Salt & Marine Chemicals Research Institute Bhavnagar India

Cristina Panisi

Department of Brain and Behavioral Sciences University of Pavia Pavia Italy

Chris Paxos

Department of Psychiatry College of Medicine and Department of Pharmacy Practice College of Pharmacy Northeast Ohio Medical University Rootstown, OH USA

Regino Perez-Polo

Department of Biochemistry and Molecular Biology University of Texas Medical Branch Galveston, TX USA

Pierluigi Politi

Department of Brain and Behavioral Sciences University of Pavia Pavia Italy

Hisham Qosa

Department of Basic Pharmaceutical Sciences School of Pharmacy University of Louisiana at Monroe Monroe, LA USA

Justine Renaud

Department of Medical Biology and Cellular Signalling Research Group University of Quebec Trois-Rivières, QC Canada xvi Contributors

Jason R. Richardson

Department of Pharmaceutical Sciences College of Pharmacy Neurodegenerative Diseases and Aging Focus Group Northeast Ohio Medical University Rootstown, OH USA

Rituparna Ghosh

Dr. B.C. Roy College of Pharmacy and Allied Health Sciences Durgapur India

Syed Ibrahim Rizvi

Department of Biochemistry University of Allahabad Allahabad India

Angélica Maria Sabogal-Guáqueta

Cellular and Molecular Neurobiology Area Neuroscience Group of Antioquia School of Medicine SIU University of Antioquia Medellín Colombia

Prabodh Sadana

Department of Pharmaceutical Sciences College of Pharmacy Neurodegenerative Diseases and Aging Focus Group Northeast Ohio Medical University Rootstown, OH USA

Rajat Sandhir Department of Biochemistry Panjab University Chandigarh India

Bibekananda Sarkar

Centre for Animal Sciences School of Basic and Applied Sciences Central University of Punjab Bathinda India

Sunil Kumar Sharma

Department of Biochemistry Panjab University Chandigarh India

Yasir Hasan Siddique

Drosophila Transgenic Laboratory Section of Genetics Department of Zoology Faculty of Life Sciences Aligarh Muslim University Aligarh India

Subhash C. Mandal

Pharmacognosy and Phytotherapy Laboratory Division of Pharmacognosy Department of Pharmaceutical Technology Jadavpur University Kolkata India

Aditya Sunkaria

Department of Biochemistry Panjab University Chandigarh India

Ameema Tariq

Atta-ur-Rahman School of Applied Biosciences National University of Sciences and Technology Islamabad Pakistan

Contributors **xvii**

Clara Tinarelli

Institute of Gerontology and Geriatrics Department of Medicine University of Perugia Perugia Italy

David Vauzour

Norwich Medical School Faculty of Medicine and Health Sciences University of East Anglia Norwich UK

Aarti Yadav

Department of Biochemistry Panjab University Chandigarh India

Kiyofumi Yamada

Department of Neuropsychopharmacology and Hospital Pharmacy Graduate School of Medicine Nagoya University Showa-ku, Nagoya Japan

Preface

The medicinal properties and health benefits of plant products (seeds, fruits, leaves, stems, and roots) are attributed to their non-nutritive bioactive components, known as "phytochemicals," which are classified into primary and secondary metabolites. Primary metabolites (carbohydrates, lipids, amino acids, and proteins) are necessary for the growth and basic metabolism of all plants. Secondary metabolites (phytochemicals), on the other hand, are not essential, but they provide vegetables, fruits, and herbs with their flavor and color. They not only play crucial roles in the well being of plants by interacting with their ecosystems, but also protect them from pathogens and absorb ultraviolet (UV), preventing DNA and photosynthetic apparatus damage. Consumption of phytochemicals by animals produces antioxidant, anti-inflammatory, antimicrobial, antitumor, analgesic, neuroprotective, and antiplatelet effects. In addition, they induce antiaging effects and improve poor blood circulation. These effects are mediated through the regulation of various receptors, transcription factors, growth factors, inflammatory cytokines, protein kinases, protein phosphatases, and other enzymes (phospholipases and cyclooxygenases). In brain, receptors, transcription factors, growth factors, and enzymes modulate the signal-transduction pathways critical in controlling synaptic plasticity and inducing neurogenesis in the hippocampus. The ability of many phytochemicals to activate the extracellular signal-regulated kinase (ERK)1/2 and protein kinase B (PKB/Akt) signaling pathways is associated with the activation of the cyclic adenosine monophosphate (cAMP) response element-binding protein (CREB), a transcription factor that plays an important role in memory formation. In recent years, the amount of research into phytochemicals has increased all over the world, and new terms such as "functional food" and "nutraceutical" have been introduced. There are several issues related to the use of phytochemicals, including concern about their dosage and activity and about the presence of contaminants.

Epidemiological studies have shown that incidences of neurological disorders among people living in Asia are lower than in the Western world. This may be due to the regular consumption of phytochemicals in the form of spices. Extensive research over the last 10 years has indicated that phytochemicals derived from various spices and oils (turmeric, black pepper, licorice, clove, ginger, garlic, green tea, and olive and flaxseed oils) target inflammatory and oxidative stress pathways and retard or delay the onset of neurological diseases. More than 7000 phytochemicals have been identified, which possess antiproliferative, anti-inflammatory, antioxidant, antiviral, and hypocholesterolemic properties. Unlike vitamins and minerals, phytochemicals are not necessary for the maintenance of cell viability, but they play a vital role in protecting neural cells from the inflammation and oxidative stress associated with normal aging and brain diseases. Although many phytochemicals present in plant foods are poorly absorbed and undergo rapid excretion, they exert anti-inflammatory, antioxidant, and anticarcinogenic effects at realistic doses. Consumption of phytochemicals may also mediate neurohormetic response through the modulation of adaptive stress-resistance genes, which are responsible for encoding protein chaperones that favor resistance to cellular stress and modulate immune function. Thus, regular consumption of phytochemicals from childhood to adulthood may reduce the risk of age-related neurological disorders.

The chemical structures of phytochemicals are often used as "privileged structures" for the creation of synthetic analogues, which have improved pharmacological activities due to their optimized bioavailability and pharmacokinetic profile. Note that most studies on phytochemicals have been performed in animal models and cell-culture systems, and it is difficult to evaluate the significance of their effect in humans.

Information on the effects of phytochemicals on human health is scattered throughout the literature in the form of original papers and reviews, but few edited books. In this book, we provide the reader with a comprehensive and cutting-edge description of the metabolism of the molecular mechanism associated with the beneficial effects of phytochemicals in age-related neurological disorders, in a manner that is useful not only to students and teachers but also to researchers and physicians. The book has 29 chapters. Chapter 1 provides an introduction to the role of phytochemicals in protecting against neuroinflammation, which is typically associated with neurodegenerative diseases. Chapter 2 deals with the protective role of flavonoids in transgenic Alzheimer's disease (AD) mouse models. Chapters 3–15 describe the beneficial effects of phytochemicals (rich in flavonoids and polyphenols) against neurological disorders in model systems. Chapter 16 discusses the use of bee products (apitherapy) for the treatment of neurological disorders. Chapter 17 elegantly describes the mechanisms underlying the beneficial actions of polyunsaturated fatty acids (PUFAs) in brain diseases. Chapters 18-20 deal with the anti-inflammatory effects of resveratrol. Chapter 21 focuses on nobiletin, a flavonoid (an O-methylated flavone) that has the ability to rescue cognitive impairment in animal models. Chapters 22-25 discuss the potential neuroprotective effects of curcumin against brain diseases. Chapter 26 discusses polyphenols and protein misfolding. Chapter 27 describes the molecular mechanisms involved in the neuroprotective action of phytochemicals. Chapter 28 focuses on nutraceuticals (a food or a part of a food that provides health benefits, including the prevention or treatment of a disease) and their effect on cognitive dysfunction. Finally, Chapter 29 provides a perspective on the importance of phytochemicals in diet and on the direction for future research in phytotherapeutics. These topics fall in a fast-paced research area related to cell death in neurological disorders, which provides opportunities for target-based therapeutic intervention using phytochemicals. This book can be used as a supplemental text for a range of phytotherapeutics courses. Clinicians and pharmacologists will find it useful in understanding the molecular aspects of phytochemicals in chronic human diseases.

We have tried to ensure uniformity of presentation, as well as a logical progression of subject from one topic to another, and our authors have provided extensive bibliographies. For the sake of simplicity and consistency, a large number of figures showing the chemical structures of phytochemicals used for the treatment of chronic diseases and signal-transduction pathways are also included. We hope that our attempt to integrate and consolidate the current knowledge on the molecular aspects of phytochemicals will provide the basis for more dramatic advances and developments in the area of the molecular mechanisms associated with the beneficial effects of phytochemicals in age-related neurological disorders.

> Tahira Farooqui Akhlaq A. Farooqui

Acknowledgments

We thank all the authors who shared their expertise by contributing chapters of a high standard, thus making our editorial task much easier. We are grateful to Justin Jeffryes, Editorial Director at Wiley-Blackwell, Health and Life Sciences, for his advice, cooperation, and understanding during compilation of this book. We are also thankful to Sumathi Elangovan, Project Editor and Jerusha Govindakrishnan, Production Editor at Wiley-Blackwell, for handling the production process in a most efficient and cooperative manner.

> Tahira Farooqui Akhlaq A. Farooqui

Use of Phytochemicals against Neuroinflammation

Wei-Yi Ong,^{1,2} Tahira Farooqui,³ Christabel Fung-Yih Ho,¹ Yee-Kong Ng,¹ and Akhlaq A. Farooqui⁴ 1

¹ Department of Anatomy, National University of Singapore, Singapore

² Neurobiology and Ageing Research Programme, National University of Singapore, Singapore

³ Department of Entomology, Ohio State University, Columbus, OH, USA

⁴ Department of Molecular and Cellular Biochemistry, Ohio State University, Columbus, OH, USA

1.1 Introduction

Neuroinflammation and oxidative stress are closely associated with the pathogenesis of neurotraumatic and neurodegenerative diseases, such as stroke and Alzheimer's disease (AD). During the inflammatory reaction, secretion of proinflammatory cytokines and chemokines amplifies and maintains inflammatory responses. It involves the enzymatic activity of cytosolic phospholipase A₂ (cPLA₂) and secretory phospholipase A₂ (sPLA₂), which release arachidonic acid from glycerophospholipids, and of cyclooxygenase (COX) and 5-lipoxygenase (5-LOX), which oxidize arachidonic acid to proinflammatory eicosanoids. This is followed by the formation of the prostaglandin D2 (PGD2) and of docosahexaenoic acid (DHA)-derived resolvins and protectins, which facilitate the resolution of inflammation. Acute neuroinflammation is a protective process that isolates the injured brain tissue from uninjured areas, destroys injured cells, and rebuilds the extracellular matrix. Without it, brain tissue would rapidly be damaged by the effects of injury and infections, including those of microbial, viral, and prion origin. Acute neuroinflammation involves the recruitment of lymphocytes, monocytes, and macrophages of the hematopoietic system and glial cells of the central nervous system (CNS). Microglia are recruited to the site of injury to protect and repair the injured tissue via the secretion of cytokines, chemokines, and lipid mediators such as resolvins and neuroprotectins, while astrocytes react by forming a glial scar. Chronic neuroinflammation, on the other hand, lingers for years, and causes damage to brain tissues. It is closely associated with the activity of microglia and astrocytes and with the assembly and activation of the inflammasome: a multiprotein oligomer consisting of caspase 1, PYCARD, NALP, and sometimes caspase 5 (also known as caspase 11 or ICH-3). Once activated, the inflammasome binds to and appositions together many p45

Neuroprotective Effects of Phytochemicals in Neurological Disorders, First Edition. Edited by Tahira Farooqui and Akhlaq A. Farooqui. © 2017 John Wiley & Sons, Inc. Published 2017 by John Wiley & Sons, Inc.

1

2 Neuroprotective Effects of Phytochemicals in Neurological Disorders

pro-caspase-1 molecules to induce their autocatalytic cleavage to p20 and p10 subunits. Caspase-1 then assembles into its active form (consisting of two heterodimers with a p20 and p10 subunit each), in order to carry out a variety of processes, including cleavage of pro-interleukin (IL)-1 β into IL-1 β , cleavage of pro-IL-18 into IL-18 to induce interferon gamma (IFN- γ) secretion, and activation of lipid biosynthesis [1]. Inflammasomes orchestrate the activation of precursors of proinflammatory caspases, which, in turn, cleave precursor forms of IL-1 β , IL-18, and IL-33 into their active forms. These lead to further stimulation of PLA₂, COX-2, and LOX; generation of eicosanoids, lysophosphatidylcholine (lyso-PtdCho), and platelet-activating factor (PAF); production of reactive oxygen species (ROS), proteinases, and complement proteins; and a potent inflammatory response. Alterations in the expression of inflammasome mediators may lead to neurodegeneration in neurotraumatic, neurodegenerative, and neuropsychiatric diseases. Based on this, it has been suggested that regulation of the inflammatory conditions [1,2].

An emerging approach to the alleviation of neuroinflammation involves the use of medicinal plants and herbs. Epidemiological studies have indicated that the incidence of neurological disorders among people living in Asia is lower than that in the Western world. This may be due to the regular consumption of phytochemicals in the form of spices. Extensive research over the last 10 years has indicated that phytochemicals derived from various spices e.g., turmeric, red pepper, black pepper, licorice, clove, ginger, garlic, coriander, cinnamon, target inflammatory and oxidative stress pathways and retard or delay the onset of neurological diseases. More than 7000 phytochemicals, which possess antiproliferative, anti-inflammatory, antiviral, and hypocholesterolemic properties, have been identified (Figure 1.1). Unlike vitamins and minerals, phytochemicals are not required for the maintenance of cell viability, but play a vital role in protecting neural cells from neuroinflammation and oxidative stress associated with aging and brain diseases. Roots, stems, leaves, fruits, and seeds contain phytochemicals such as terpenoids, phenolic compounds, glucosinolates, betalains, and chlorophylls. Although many phytochemicals in plant foods are poorly absorbed and undergo rapid excretion, they exert anti-inflammatory, antioxidant, and anticarcinogenic effects at realistic doses. The effects of phytochemicals are mediated by their ability to counteract, reduce, and repair damage resulting from oxidative stress and



Figure 1.1 Effect of phytochemicals on various cellular activities.

neuroinflammation – processes that are modulated by the transcription factor, nuclear factor kappa B (NF- κ B). Phytochemicals also stimulate the synthesis of adaptive enzymes and proteins that favor resistance to cellular stress [3].

1.2 Mechanism of Action of Phytochemicals

Plants and phytochemicals produce their beneficial effects not only through modulation of enzyme activities and regulation of gene expression, but also via the stimulation of adaptive cellular stress response pathways that protect cells against a variety of adverse conditions. Phytochemicals bind to neuronal cell-membrane or nuclear receptors as elective ligands and have signaling effects at concentrations much lower than is required for effective antioxidant activity [4]. They act on the NF-κB pathway to inhibit inflammation. NF- κ B is predominantly localized in the cytoplasm in a complexed form that is inactive, but during oxidative stress it is released from the NF- κ B-I κ B α complex and migrates to the nucleus, where it initiates the transcription of a number of proinflammatory enzymes, including sPLA₂, COX-2, NADPH oxidase and inducible nitric oxide synthase (iNOS), as well as proinflammatory cytokines (tumor necrosis factor alpha (TNF- α), IL-1 β , and IL-6). The latter stimulate the activities of PLA₂ and sphingomyelinases through a feedback loop involving cytokine-mediated phosphorylation. Other potential mechanisms through which NF-κB induces neuronal death include the induction of death proteins and an aborted attempt to re-enter the cell cycle. Phytochemicals such as curcumin, resveratrol, Ginkgo biloba (GB) retard inflammation by preventing the migration of NF- κ B into the nucleus. In addition, many phytochemicals block the activation of NF-kB by inhibiting a protein kinase. In vitro studies indicate that phytochemicals inhibit both serine/threonine protein kinase and protein tyrosine kinase, supporting the view that phytochemicals may inhibit I κ B kinase β (IKK β) in the cytoplasm and nucleus, leading to a reduction in NF-κB activity. Phytochemicals have also been reported to modulate age-related decline in memory by upregulating signaling pathways that control synaptic plasticity. They activate both the extracellular signalregulated kinase (ERK) 1/2 and protein kinase B (PKB)/Akt signaling pathways and cyclic adenosine monophosphate (cAMP) response element-binding protein (CREB), a transcription factor that upregulates the expression of several neurotrophins that facilitate memory formation [5,6].

An important cellular antioxidant response that underlies the action of many phytochemicals is induction of antioxidative and anti-inflammatory enzymes through the cytoplasmic oxidative stress system (nuclear factor erythroid 2-related factor 2 (Nrf2)– kelch-like erythroid Cap'n'Collar homologue-associated protein 1 (Keap1)) (Figure 1.2) [7]. Under physiological conditions, Keap1 keeps the Nrf2 transcription factor in the cytoplasm, allowing it to be ubiquitinated and degraded by proteasomes, thus maintaining Nrf2 at low levels. This prevents Nrf2 from mediating the constitutive expression of its downstream genes. When cells are exposed to oxidative stress, a signal involving phosphorylation and/or redox modification of critical cysteine residues in Keap1 blocks the enzymatic activity of the Keap1–Cul3–Rbx1 E3 ubiquitin ligase complex, leading to a decrease in Nrf2 ubiquitination and degradation. As a result, free Nrf2 translocates into the nucleus, where it – along with other transcription factors (e.g., sMaf, ATF4, JunD, PMF-1) – transactivates the antioxidant response elements (AREs)



Figure 1.2 Hypothetical diagram showing the effects of phytochemicals on signal transduction processes in the brain. AA, arachidonic acid; COX-2, cyclooxygenase 2; cPLA₂, cytosolic phospholipase A₂; HO-1, hemeoxygenase 1; HSP, heat-shock protein; IL-1 β , interleukin 1 β ; IL-6, interleukin 6; iNOS, inducible nitric oxide synthase; Keap1, kelch-like erythroid Cap'n'Collar homologue-associated protein 1; LOX, lipoxygenase; lyso-PtdCho, lyso-phosphatidylcholine; NF- κ B, nuclear factor kappa B; Nrf2, nuclear factor erythroid 2-related factor 2; PAF, platelet-activating factor; PM, plasma membrane; PtdCho, phosphatidylcholine; QR, quinine oxidoreductase; ROS, reactive oxygen species; SOD, superoxide dismutase; sPLA₂, secretory phospholipase A₂; TNF- α , tumor necrosis factor alpha; γ -GCL, gamma glutamylcystein ligase. (*See insert for color representation of the figure*.)

of many cytoprotective genes. Microarray and biochemical analyses reveal the coordinated upregulation of several enzymes, such as HO-1, catalase, SOD, epoxide hydrolase, UDP-glucuronosyltransferases, and thioredoxin. In addition, Nrf2 induces the expression of enzymes related to glutathione biosynthesis and function (e.g., xCTcystine antiporter, gamma-glutamylcysteine synthetase, reduced glutathione (GSH) synthase, glutathione S-transferase (GST), glutathione reductase (GR)), leading to an increase in intracellular GSH and a decrease in oxidative stress. Upon recovery of cellular redox status, Keap1 travels into the nucleus and facilitates the dissociation of Nrf2 from ARE. Subsequently, the Nrf2–Keap1 complex is exported out of the nucleus by the nuclear export sequence in Keap1. Once in the cytoplasm, the Nrf2–Keap1 complex associates with the Cul3–Rbx1 core ubiquitin machinery, leading to degradation of Nrf2 and termination of the Nrf2/ARE signaling pathway [7].

Phytochemicals may also act through oxidant-mediated neural cell survival signaling pathways, together with histone deacetylases of the sirtuin family (sirtuin–FOXO pathway) and chaperones such as the heat-shock proteins (HSPs), antioxidant enzymes (SODs and glutathione peroxidase (GPx)), and growth factors (e.g., insulinlike growth factor (IGF), brain-derived neurotrophic factor (BDNF)) [8–10]. Low levels of phytochemicals crossing the blood–brain barrier (BBB) cause mild cellular inflammation and oxidative stress, involving the generation of low levels of ROS, which results in activation of transcription factors and synthesis of HSPs promoting the production of anti-inflammatory cytokines. In this scenario, responses to HSPs are considered an attempt to correct the inflammatory condition. The highly integrated and regulated processes are controlled by redox-sensitive genes called "vitagenes," which code for HSPs, thioredoxin, and sirtuin protein systems and modulate a complex network of intracellular signaling pathways for the preservation of cellular homeostasis.

A potential mechanism through which phytochemicals could exert their effect is via the inflammasome. *In vitro* studies have indicated that phytochemicals partially inhibit the release of TNF- α , IL-1 β , and IL-6 in cultured neural cells [3]. TNF- α and IL-1 β activate isoforms of PLA₂ and sphingomyelinases. It appears that phytochemicals prevent the formation of inflammasome by inhibiting the activities of NLRP3, NLRC4, AIM2, NLRP6, caspase-1, PLA₂, and sphingomyelinases, although more work needs to be carried out in this area.

1.3 Bioavailability of Phytochemicals

Bioavailability represents the fraction of an orally ingested or administered compound in food, beverages, or supplements that reaches the systemic circulation. The bioavailability of most phytochemicals in human tissues is very poor. Following oral administration, most phytochemicals are absorbed and metabolized to form glucuronide and sulfate conjugates, which are excreted in the urine [3,11]. The bioavailability of most phytochemicals in peripheral organs is higher than that in the brain as a result of the presence of the BBB. In order to enter the brain, a phytochemical must either be highly lipid-soluble or be subjected to uptake transport processes through adenosine triphosphate (ATP)-binding cassette (ABC) transporters. Many approaches have been taken in

6 Neuroprotective Effects of Phytochemicals in Neurological Disorders

an attempt to improve the bioavailability of phytochemicals, including the use of adjuvants that interfere with glucuronidation, the preparation of phytochemical liposomes and nanoparticles, the use of phytochemical–phospholipid conjugates, and the use of structural analogs of phytochemicals. These approaches have generally allowed phytochemicals to cross the BBB more effectively [12–14].

Green tea contains catechin flavonoid polyphenols. Catechin monomers can be easily absorbed through the gut, whereas large molecular-weight catechins, such as (–)-epigallocatechin-3-gallate (EGCG) (Figure 1.3), are poorly absorbed. Green-tea catechins undergo three degradation processes: decomposition to smaller molecules, polymerization to larger molecules, and oxidation to oxidized molecules under natural conditions. The digestive tract plays an important role in the metabolism and bioavailability of green-tea components before they enter the liver. Green-tea catechins and their metabolites formed in the small intestine are transported back into the intestinal lumen, where they reach the large intestine and are broken down to small phenolic acids and valerolactones by resident microflora. These metabolites are either reabsorbed or excreted in the feces [15].

The bioavailability of flavonoid polyphenols in berries is very low, and information on the molecular mechanisms of their action is still poorly understood. Dietary flavonoids enter the gastrointestinal tract in the form of esters and glycosides, which are not easily absorbed. Conversion of esters and glycosides into aglycones results in better bioavailability, because aglycones are lipophilic and more permeable across the cell membrane than the parent glycosides, and are more efficiently absorbed across the



Figure 1.3 Chemical structures of phytochemicals.

gastrointestinal tract wall [16,17]. The conversion of glycosides into aglycones mainly occurs in the acidic environment of the stomach. Absorption of flavonoids by intestinal epithelial cells is accompanied by their extensive biotransformation, with the generation of different conjugated products (e.g., glucuronides, sulfates, O-methylated derivatives), first in the intestine and then in the liver, where conjugates are secreted into the bile. Favorable absorption across the gastrointestinal tract does not always result in improved bioavailability. One possible approach to improving the bioavailability of phytochemicals in the brain is the use of nanolipidic particles [18].

Resveratrol (3,5,4'-trihydroxy-trans-stilbene) (Figure 1.3) is a member of the stilbenoid family of polyphenols. It is found in grapes, mulberries, peanuts, and other plants and food products, including raspberries, blueberries, Scots pine, Eastern white pine, and knotweed. Although transresveratrol is rapidly absorbed and distributed through the bloodstream in animals and humans, its bioavailability is low due to its rapid metabolism and elimination in the urine [19,20]; extensive metabolism in the intestine and liver results in an oral bioavailability of less than 1%. Resveratrol is well tolerated and metabolized through glucuronidation or sulfation reactions in the intestine and liver. The major glucuronidation derivatives of resveratrol are transresveratrol 3-O-glucuronide and transresveratrol-O-glucuronide, whereas the sulfated derivative is transresveratrol-3-O-sulfate. Kinetic analysis of resveratrol metabolism indicates that glucuronidation is favored over sulfation in the liver. *In vivo* studies indicate that free transresveratrol levels in the plasma are very low and short-lived. Intravenous administration of 15 mg/kg in rats results in a wide distribution of resveratrol in various tissues after 90 minutes. The highest concentrations are found in the kidney and lowest in the brain [21].

Curcumin is a member of the curcuminoid family of polyphenols (Figure 1.3). It has poor bioavailability, due either to (i) its poor absorption, (ii) its rapid metabolism, or (iii) its rapid systemic elimination and short biological half-life. In rodents, curcumin undergoes rapid metabolism by conjugation, reduction, and removal after oral dosing. Very little information is available on the pharmacokinetics of curcumin in humans. Phase I and II clinical trials have been performed for up to 4 months at several doses (500, 1000, 2000, 4000, 8000, and 12000 mg/day) in patients with advanced colorectal cancer, without any toxicity [22,23]. The serum concentration of curcumin reaches a maximum at 1-2 hours after oral intake and gradually declines within 12 hours. Currently, an upper level of toxicity has not been established for curcumin. Studies have reported that a daily dose of as high as 12g is safe and tolerable in humans, with few, mild side effects. Hybrid compounds of curcumin and melatonin have been designed, synthesized, and characterized, and one of these has been shown to cross the BBB and deliver a sufficient amount to brain tissue following oral administration. Results suggest the hybridization approach is an efficient strategy for identifying novel scaffolds with the desired pharmacology for use as neuroprotectants. In order to increase curcumin's bioavailability, a polymeric nanoparticle encapsulated curcumin (NanoCurc) has been formulated. NanoCurc injection at a dose of 25 mg/kg twice daily in mice results in significant curcumin levels in the brain $(0.32 \mu g/g)$ [24].

The bioavailability of flavonoids of GB extract EGb 761 is generally low due to limited absorption and rapid elimination [25]. Unabsorbed flavonoids that reach the colon are metabolized by colon microflora and absorbed. The flavonoids then reach the liver, where they are metabolized to conjugated derivatives. The bioavailability of EGb 761 has been studied in rats and humans. Oral administration or injection of acute and subacute doses of EGb 761 in rats results in the distribution of GB components in various

8

tissues and plasma that follows linear pharmacokinetics in rats and humans. In human volunteers, oral intake of EGb 761 (120 mg) results in bilobalide plasma levels of $0.05-0.15\,\mu$ M [26,27]. In rats, oral administration of EGb 761 or pure bilobalide produces dose-dependent increases in bilobalide plasma levels of from 0.5 to 7.5 mM. It is increasingly evident that components of EGb 761 can cross the BBB and reach low micromolar concentrations in the brain. This allows efficient interaction with target molecules such as neurotransmitter receptors [28,29].

Studies on the bioavailability of a component of garlic, S-allyl cysteine (SAC), which belongs to the organosulfur family of glycosinolates, indicate that oral consumption results in rapid absorption in the gastrointestinal tract and distribution in the plasma, liver, and kidney of rats, mice, and dogs [30]. The bioavailability of SAC is about 100.0% in mice, 98.2% in rats, and 87.2% in dogs. N-acetyltransferases transform SAC into N-Acetyl-SAC, which can be detected in the urine of dogs and humans. Other oil-soluble organosulfur compounds of garlic, such as allicin, sulfides, ajoene, and vinyldithiins, are not found in the blood or urine even after consumption of large amounts of garlic. Incubation of allicin with liver homogenates results in its very rapid disappearance. No allicin was detected in either blood or urine 1–24 hours after ingestion of 25 g of raw garlic (approximately 90 mg allicin). In any case, allicin quickly disappears from the blood within a short period after ingestion, and its decomposition products diallyl sulfide and allylmercaptan are found in the blood [31].

1.4 Plants Effective against Neuroinflammation

Large number of plants around the world have been found to reduce inflammatory responses or stimulate antioxidant defenses, both in microglial cells and neurons *in vitro*, and in animal models of neurological diseases. On compiling literature reports of plants with antineuroinflammatory properties, we discovered that they are not randomly distributed throughout the plant kingdom but are **concentrated in a small number of orders**, especially *Fabales*, *Lamiales*, *Rosales*, *Apiales* and *Sapindales* (Table 1.1). These same orders of plants are associated with food allergy [32]. We hypothesize that plants that are useful against neuroinflammation are also those that are mildly proinflammatory or immunogenic.

Very low levels of phytochemicals that enter the brain due to limited absorption through the gastrointestinal tract and very limited passage across the BBB may cause a mild neuroinflammation and trigger the activation of transcription factors and the synthesis of HSPs, leading to an anti-inflammatory response. This may be a form of "inflammatory" hormesis (a process whereby a specific phytochemical induces biologically opposite effects at different doses).

Knowledge of the orders of plants that are most likely to be effective against neuroinflammation may help in the future discovery of novel phytochemicals. Low levels of phytochemicals are closely associated with the adaptive stress response, which confers resistance to severe inflammation and stress, through the activation of the Nrf2 pathway and antioxidant/drug-metabolizing enzymes, and through the generation of low levels of lipid mediators. These mediators maintain the cellular milieu and transfer messages between subcellular organelles, thereby inducing physiological functions (e.g., bioenergetics, growth, proliferation, remodeling). These

Common (Chinese) name	Order	Family	Genus	Species
Chinese ginseng/Korean ginseng	Apiales	Araliaceae	Panax	ginseng
Siberian ginseng	Apiales	Araliaceae	Eleutherococcus/Acanthopanax	senticosus
Asiatic/Indian pennywort (ji xue cao)	Apiales	Apiaceae	Centella	asiatica
Angelica dahuricae (bai zhi)	Apiales	Apiaceae	Angelica	dahuricae
Açaí palm	Arecales	Arecaceae	Euterpe	oleracea
African oil palm	Arecales	Arecaceae	Elaeis	guineensis
Garlic	Asparagales	Amaryllidaceae	Allium	sativum
Long yellow day lily (huang hua cai)	Asparagales	Xanthorrhoeaceae	Hemerocallis	citrina
Milk thistle	Asterales	Asteraceae	Silybum	marianum
Chinese mugwort (ai cao)	Asterales	Asteraceae	Artemisia	argyi
Lavender cotton	Asterales	Asteraceae	Achillea	fragrantissima
Coltsfoot	Asterales	Asteraceae	Tussilago	farfara
Thunder god vine (lei gong teng)	Celastrales	Celastraceae	Tripterygium	wilfordii
Bitter melon/bitter gourd	Cucurbitales	Cucurbitaceae	Momordica	charantia
Valeriana amurensis Smir. ex Kom. (hei shui xie cao)	Dipsacales	Caprifoliaceae	Valeriana	amurensis
Green tea (cha hua)	Ericales	Theaceae	Camellia	sinensis
Berry	Ericales	Ericaceae	Vaccinium	I
Walnut	Fagales	Juglandaceae	Juglans	regia
Yuan zhi	Fabales	Polygalaceae	Polygala	tenuifolia
Velvet bean	Fabales	Fabaceae	Mucuna	pruriens
Soybean	Fabales	Fabaceae	Glycine	тах
Ku shen	Fabales	Fabaceae	Sophora	flavescens
				(Continued)

Table 1.1 Classification of plants with potential against neuroinflammation.

Table 1.1 (Continued)

Common (Chinese) name	Order	Family	Genus	Species
Kudzu (gan ge teng)	Fabales	Fabaceae	Pueraria	thomsonii
Squirrel's claws/kuku tupai (hua nan yun shi)	Fabales	Fabaceae	Caesalpinia	crista
Jatobá-do-cerrado	Fabales	Fabaceae	Нутепаеа	stigonocarpa
Ginkgo biloba (yin xing)	Ginkgoales	Ginkgoaceae	Ginkgo	biloba
Coffee	Gentianales	Rubiaceae	Coffea	arabica
Cryptolepis sanguinolenta	Gentianales	Apocynaceae	Cryptolepis	sanguinolenta
Olive	Lamiales	Oleaceae	Olea	europaea
Forsythia	Lamiales	Oleaceae	Forsythia	koreana
Brahmi/Indian pennywort	Lamiales	Plantaginaceae	Bacopa	топпіегі
Sesame	Lamiales	Pedaliaceae	Sesamum	indicum
Devil's claw	Lamiales	Pedaliaceae	Harpagophytum	procumbens
Baikal skullcap (huang qin)	Lamiales	Lamiaceae	Scutellaria	baicalensis
Greek sage	Lamiales	Lamiaceae	Salvia	fruticosa
Cinnamon	Laurales	Lauraceae	Сіппатотит	cassia
Magnolia bark (hou po)	Magnoliales	Magnoliaceae	Magnolia	officinalis
Willow bark	Malpighiales	Salicaceae	Salix	I
Wild viola (bai hua di ding)	Malpighiales	Violaceae	Viola	patrinii
Cnestis ferruginea	Oxalidales	Connaraceae	Cnestis	ferruginea
Maritime pine/cluster pine	Pinales	Pinaceae	Pinus	pinaster
Thuja orientalis	Pinales	Cupressaceae	Thuja	orientalis
Black pepper	Piperales	Piperaceae	Piper	nigrum
Kava/kava-kava	Piperales	Piperaceae	Piper	methysticum

Black seed	Ranunculales	Ranunculaceae	Nigella	sativa
Cannabis/marijuana	Rosales	Cannabaceae	Cannabis	sativa
White mulberry	Rosales	Moraceae	Morus	alba
Cherokee rose (jin ying zi)	Rosales	Rosaceae	Rosa	laevigata
Nettle	Rosales	Urticaceae	Urtica	dioica
Hawthorn (shan zha)	Rosales	Rosaceae	Crataegus	oxyacantha
Myrrh	Sapindales	Burseraceae	Commiphora	erythraea
Mango	Sapindales	Anacardiaceae	Mangifera	indica
Mastic tree	Sapindales	Anacardiaceae	Pistacia	lentiscus
Chinese toon (xiang chun)	Sapindales	Meliaceae	Тоопа	sinensis
Mother-of-pearl-plant/ghost plant	Saxifragales	Crassulaceae	Graptopetalum	paraguayense
Chinese peony (bai shao)	Saxifragales	Paeoniaceae	Paeonia	lactiflora
Artic root/golden root (mei gui hong jing tian)	Saxifragales	Crassulaceae	Rhodiola	rosea
Wolfberry/goji berry (ning xia gou qi)	Solanales	Solanaceae	Lycium	barbarum
Sweet potato	Solanales	Convolvulaceae	Ipomoea	batatas
Grapevine	Vitales	Vitaceae	Vitis	vinifera
Turmeric	Zingiberales	Zingiberaceae	Curcuma	longa
Ginger	Zingiberales	Zingiberaceae	Zingiber	officinale
Brown algae	Fucales	Sargassaceae	Myagropsis	myagroides
Button mushroom/portobello mushroom	Agaricales	Agaricaceae	Agaricus	bisporus
Lingzhi mushroom (ling zhi)	Polyporales	Ganodermataceae	Ganoderma	lucidum

12 Neuroprotective Effects of Phytochemicals in Neurological Disorders

processes are partially controlled by the redox-sensitive vitagenes (see Section 1.2). At high doses, the same phytochemicals may elicit toxic effects through the activation of NF- κ B, increased expression of cytokines and chemokines, activation of PLA₂s and COX, and generation of high levels of ROS and inflammatory eicosanoids. The amount of fruits and vegetables normally consumed by humans falls within the subtoxic stimulatory dose range of concentrations. However, some plants and fungi produce and concentrate toxins in amounts sufficient to cause illness or death in humans [33,34].

1.4.1 Order: Apiales

1.4.1.1 Family: Araliaceae, Genus: Panax, Species: ginseng

Panax ginseng is widely used as a health tonic. Ginsenoside remodulated phospho-p38, iNOS, and COX-2 signaling pathways in lipopolysaccharide (LPS)-stimulated BV-2 microglial cells. Ginsenoside Rg1 (50, 100 or 150 μ M) decreased IL-1 β , IL-8, and TNF- α levels in amyloid beta (A β)40-treated THP-1 monocytes [35].

1.4.1.2 Family: Araliaceae, Genus: Eleutherococcus/Acanthopanax,

Species: senticosus

Acanthopanax senticosus (Siberian ginseng) extracts modulated nitric oxide (NO)/ROS production and induced translocation of Nrf2 to increase hemeoxygenase 1 (HO-1) expression in LPS-stimulated BV-2 microglial cells. Extracts also reduced infarct volume after cerebral ischemia in rats [36].

1.4.1.3 Family: Apiaceae, Genus: Centella, Species: asiatica

Centella asiatica is a medicinal herb commonly used in Ayurveda and traditional Chinese medicine. Asiaticoside, a triterpenoid isolated from *Centella asiatica*, modulated COX-2 expression in the brains of LPS-stimulated animals. Phenolic acids identified in *Centella asiatica* extracts also inhibited 5-LOX and were shown to have antioxidant and anti-inflammatory effects [37].

1.4.1.4 Family: Apiaceae, Genus: Angelica, Species: dahuricae

Angelica dahuricae radix extract modulated TNF- α , IL-1 β , IL-6, iNOS, COX-2, and ROS in LPS-stimulated BV-2 microglial cells. Oral administration of the extract suppressed caspase-3 activation and apoptotic cell death of neurons and oligodendrocytes, and improved function after spinal cord injury (SCI) [38].

1.4.2 Order: Arecales

1.4.2.1 Family: Arecaceae, Genus: Euterpe, Species: oleracea

Euterpe oleracea Mart (açaí) fruit pulp has a high content of polyphenols. Açaí fractions modulated iNOS, COX-2, p38 mitogen-activated protein kinase (MAPK), TNF- α , and NF- κ B expression in BV-2 microglial cells [39].

1.4.2.2 Family: Arecaceae, Genus: Elaeis, Species: guineensis

The oil palm (*Elaeis guineensis*) is an abundant source of water-soluble phenolics. In mice treated with oil palm phenolics, genes involved in brain development and activity were upregulated, while those involved in inflammation were downregulated [40].

1.4.3 Order: Asparagales

1.4.3.1 Family: Amaryllidaceae, Genus: Allium, Species: sativum

Intraperitoneal administration of a component of "Aged Garlic Extract" (AGE), SAC, modulated oxidative stress and decreased infarct volume after cerebral ischemia in rats. SAC also inhibited cell signaling pathways involved in synaptic degeneration and neuroinflammation in AD [41].

1.4.3.2 Family: Xanthorrhoeaceae, Genus: Hemerocallis, Species: citrina

Hemerocallis citrina, a traditional herbal medicine, has been used for the improvement of behavioral and emotional status in East Asian countries. It modulated IL-1 β , IL-6, and TNF- α expression and indoleamine-2,3-dioxygenase (IDO) activity in the frontal cortex and hippocampus of rats exposed to chronic unpredictable mild stress [42].

1.4.4 Order: Asterales

1.4.4.1 Family: Asteraceae, Genus: Silybum, Species: marianum

Silybum marianum (milk thistle) contains silymarin, which modulated inhibitor κ B-alpha (I κ B- α) degradation and NF- κ B nuclear translocation and reduced infarct volume after cerebral ischemia in rats. Silymarin has also been proposed as a neuroprotective agent in AD and Parkinson's disease (PD) [43].

1.4.4.2 Family: Asteraceae, Genus: Artemisia, Species: argyi

Artemisia argyi is a herbaceous perennial plant native to China, Japan, and Eastern Siberia. An extract modulated NO, prostaglandin E2 (PGE2), TNF- α , iNOS, COX-2, IL-1 β , granulocyte-macrophage colony-stimulating factor, and macrophage inflammatory protein-1 α levels in LPS-stimulated BV-2 microglial cells [44].

1.4.4.3 Family: Asteraceae, Genus: Achillea, Species: fragrantissima

Achillea fragrantissima is a desert plant used in traditional medicine. An extract modulated NO, IL-1 β , TNF- α , matrix metallopeptidase 9, COX-2, and iNOS levels in LPS-stimulated primary microglial cells [45].

1.4.4.4 Family: Asteraceae, Genus: Tussilago, Species: farfara

Tussilago farfara L. (Compositae) flower buds are used in traditional oriental medicine for the treatment of bronchitis and asthma. Extracts modulated arachidonic acid metabolism, neuronal injury induced by an NO generator spermine (NONOate), and A β -induced neuronal injury [46].

1.4.5 Order: Celastrales

1.4.5.1 Family: Celastraceae, Genus: Tripterygium, Species: wilfordii

Triptolide is one of the major active components of the Chinese herb *Tripterygium* wilfordii Hook F, which has potent anti-inflammatory and immunosuppressive characteristics. Extracts modulated TNF- α and NO expression and reduced injury in LPS-stimulated primary mesencephalic neurons [47].

1.4.6 Order: Cucurbitales

1.4.6.1 Family: Cucurbitaceae, Genus: Momordica, Species: charantia

Momordica charantia, often called bitter melon or bitter gourd, is reported to have anti-inflammatory properties. Extracts modulated expression of neuroinflammatory markers NF- κ B, IL-16, IL-22, and IL-17R in the brains of high-fat diet-treated mice [48].

1.4.7 Order: Dipsacales

1.4.7.1 Family: Caprifoliaceae, Genus: Valeriana, Species: amurensis

Valeriana amurensis extracts modulated iNOS, COX-2, and I κ B- α levels, reduced neuronal injury, and improved spatial exploratory activity in a rat model of AD [49].

1.4.8 Order: Ericales

1.4.8.1 Family: Theaceae, Genus: Camellia, Species: sinensis

The beneficial effects of green tea result from interactions between green-tea catechins and cellular proteins. Teasaponin, a tea extract, has also been shown to have anti-inflammatory effects [50].

1.4.8.2 Family: Theaceae, Genus: Vaccinium

Fruits such as blueberries contain flavonoids, which have anti-inflammatory and antioxidant effects. Blueberry polyphenol supplementation for 8 weeks modulated IL-1 β , TNF- α , and NF- κ B levels, and reduced learning impairment after kainate-induced excitotoxic injury in rats [51].

1.4.9 Order: Fagales

1.4.9.1 Family: Juglandaceae, Genus: Juglans, Species: regia

Extracts of *Juglans regia* (English walnut) modulated NO and iNOS levels in LPSstimulated microglial cells. Polyphenolic compounds in walnuts also reduced oxidant and inflammatory load on brain cells, and enhanced sequestration of toxic protein aggregates [52].

1.4.10 Order: Fabales

1.4.10.1 Family: Polygalaceae, Genus: Polygala, Species: tenuifolia

Polygala tenuifolia is a herb used in traditional Chinese medicine. Water extract of *Polygala tenuifolia* root modulated NO, PGE2, iNOS, COX-2, IL-1 β , and TNF- α levels in LPS-stimulated BV-2 microglial cells [53].

1.4.10.2 Family: Fabaceae, Genus: Mucuna, Species: pruriens

Mucuna pruriens, a leguminous plant, is used as an anti-inflammatory drug in Ayurveda. Ethanolic extracts downregulated NO production, neuroinflammation, and microglial activation and modulated loss of TH-positive cells in the 1-methyl-4-phenyl-1,2,3,6-tetrahydropyridine (MPTP)-treated mouse model of PD [54].

1.4.10.3 Family: Fabaceae, Genus: *Glycine*, Species: *max*

Soybean isoflavones modulated inflammatory cytokines and A β 42-induced upregulation of toll-like receptor 4 (TLR4) and NF- κ B p65 mRNA in rats [55].

1.4.10.4 Family: Fabaceae, Genus: Sophora, Species: flavescens

Sophora flavescens Ait is a herb used in traditional Chinese medicine. One of its components, oxymatrine, modulated TLR4, NF- κ B, TNF- α , IL-1 β , IL-6, 12/15-LOX, phospho-p38 MAPK, and cPLA₂ levels and reduced neuronal injury after intracerebral hemorrhage in rats [56].

1.4.10.5 Family: Fabaceae, Genus: Pueraria, Species: thomsonii

Isoflavones of *Pueraria* flowers (genistein, tectorigenin, and irisolidone) inhibited NO levels in LPS-stimulated primary microglial cells [57].

1.4.10.6 Family: Fabaceae, Genus: Caesalpinia, Species: crista

Caesalpinia crista leaf extracts showed antioxidant effects and suppressed 5-LOX in polymorphonuclear leukocytes [37].

1.4.10.7 Family: Fabaceae, Genus: Hymenaea, Species: stigonocarpa

Hymenaea stigonocarpa extracts showed protective effects following 2,4,6-trinitrobenzenesulfonic acid-induced colon damage and modulated lipid peroxidation in rat brain membranes [58].

1.4.10.8 Family: Fabaceae, Genus: Sutherlandia, Species: frutescens

Sutherlandia frutescens is native to dry parts of southern Africa. It has been promoted as useful to people with HIV/AIDS, but there is no evidence of benefit, and it may interact adversely with antiretroviral drugs. *Sutherlandia* has been shown to have anti-neuroinflammatory properties in preliminary studies (W. Folk, unpublished).

1.4.11 Order: Ginkgoales

1.4.11.1 Family: Ginkgoaceae, Genus: Ginkgo, Species: biloba

GB is a unique species of tree, with no living relatives. The active compounds of the GB extract EGb 761 interact with gamma-aminobutyric acid (GABA) and glycine receptors on neurons that play an important role in memory formation, consolidation, and cognition. EGb 761 also enhanced cholinergic processes in the cerebral cortex [59,60].

1.4.12 Order: Gentianales

1.4.12.1 Family: Rubiaceae, Genus: Coffea, Species: arabica

A component of coffee, eicosanoyl-5-hydroxytryptamideeicosanoyl-5-hydroxytryptamide, attenuated neuroinflammatory response to MPTP, and improved neuronal integrity in the α -synuclein transgenic mouse model of PD [61].

1.4.12.2 Family: Apocynaceae, Genus: Cryptolepis, Species: sanguinolenta

Cryptolepis sanguinolenta extracts modulated TNF- α , IL-6, and PGE2 levels and suppressed NF- κ B p65 nuclear translocation in IL-1 β -stimulated SK-N-SH neuroblastoma cells [62].

1.4.13 Order: Lamiales

1.4.13.1 Family: Oleaceae, Genus: Olea, Species: europaea

Olive oil contains phenolic compounds that are well-known antioxidants. Long-term consumption of olive oil increased the proportion of monounsaturated fatty acids (particularly oleic acid) and reduced the level of arachidonic acid, suggesting its potential in modulating the production of proinflammatory eicosanoids [63].

1.4.13.2 Family: Oleaceae, Genus: Forsythia, Species: koreana

Forsythia is a genus of flowering plants in the olive family. Pinoresinol isolated from the fruits of *Forsythia koreana* Nakai modulated NO, PGE2, TNF- α , IL-1 β , and IL-6 levels in LPS-stimulated primary microglial cells [64].

1.4.13.3 Family: Plantaginaceae, Genus: Bacopa, Species: monnieri

Bacopa monnieri (Indian pennywort or "Brahmi") is an important medicinal herb used in Ayurveda. Bacosides modulated proinflammatory cytokines and iNOS levels in aged rat brains. Extracts also restored Nrf2 and HO-1 expression, and improved memory dysfunction after okadaic acid treatment in rats [65].

1.4.13.4 Family: Pedaliaceae, Genus: Sesamum, Species: indicum

Sesamin, a constituent of sesame seeds, modulated extracellular signal, regulated kinase (ERK)1/2, p38 MAPK, caspase-3, and COX-2 expression in PC12 cells and BV-2 microglial cells, and reduced damage after kainate-induced excitotoxic injury in mice [66].

1.4.13.5 Family: Pedaliaceae, Genus: Harpagophytum, Species: procumbens

Harpagophytum procumbens (devil's claw) is a plant of the sesame family that is used in inflammatory diseases. Ethyl acetate extracts of *Harpagophytum procumbens* decreased lipid peroxidation and cellular damage in rat brain slices [67].

1.4.13.6 Family: Lamiaceae, Genus: Scutellaria, Species: baicalens

Scutellaria baicalens Georgia and its constituents are reported to have antioxidative and anti-inflammatory properties. Extracts modulated COX-2, iNOS, PGE2, and NO levels in LPS-stimulated Raw 264.7 and BV-2 microglial cells, and improved cognition in mice [68].

1.4.13.7 Family: Lamiaceae, Genus: Salvia, Species: fruticosa

Salvia fruticosa (Greek sage) is a perennial shrub in the eastern Mediterranean. Extracts modulated acetylcholinesterase (AChE) activity and C-reactive protein (CRP), NF- κ B, and monocyte chemoattractant protein 1 (MCP-1) levels in the AlCl₃-induced rat model of AD [69].

1.4.13.8 Family: Acanthaceae, Genus: Clinacanthus, Species: nutans

Clinacanthus nutans Lindau (Sabah snake grass) leaves have been used in traditional medicine to treat snake bite. They are also used in the treatment of cancer and of kidney failure. Our recent studies have shown *C. nutans* leaves have the ability to inhibit cPLA₂ expression in SH-SY5Y cells.

1.4.14 Order: Laurales

1.4.14.1 Family: Lauraceae, Genus: Cinnamomum, Species: cassia

Cinnamomum cassia (cinnamon) extracts modulated NO, IL-1 β , IL-6, and TNF- α levels in LPS-stimulated BV-2 microglial cells. Cinnamon's constituent, 2'-hydroxycinnamaldehyde (HCA), or its derivative, 2'-benzoyloxycinnamaldehyde (BCA), also modulated NO and TNF- α levels in LPS-stimulated microglial cells [70].

1.4.15 Order: Magnoliales

1.4.15.1 Family: Magnoliaceae, Genus: Magnolia, Species: officinalis

Oral administration of 4-O-methylhonokiol from Magnolia bark in drinking water for 12 weeks modulated β -secretase activity, A β deposition, oxidative lipid and protein damage levels, activation of glial cells, and memory impairment in the Tg2576 mouse model of AD [71]. *Magnolia officinalis* ethanol extract reduced amyloidogenesis and memory impairment in the LPS-induced mouse model of AD [72].

1.4.16 Order: Malpighiales

1.4.16.1 Family: Salicaceae, Genus: Salix

Willow bark contains salicin, which modulated immune activation and had a positive effect on the forced swim test (FST), suggesting antidepressant effects in rats [73].

1.4.16.2 Family: Violaceae, Genus: Viola, Species: patrinii

Viola patrinii extract modulated iNOS, COX-2, TNF- α , and IL-1 β levels and upregulated Nrf2-dependent expression of HO-1 in hippocampal HT22 cells and BV-2 microglial cells [74].

1.4.17 Order: Oxalidales

1.4.17.1 Family: Connaraceae, Genus: Cnestis, Species: ferruginea

Amentoflavone isolated from *Cnestis ferruginea* modulated ROS, malondialdehyde (MDA), and TNF- α levels in LPS-stimulated C6 astrocytoma cells, and reduced nociceptive responses in the carrageenan-injection mouse model of inflammatory pain [75].

1.4.18 Order: Pinales

1.4.18.1 Family: Pinaceae, Genus: Pinus, Species: pinaster

Pinus pinaster (maritime pine) is native to the western and southwestern Mediterranean. An extract from the bark, Pycnogenol, modulated TNF- α and IL-1 β levels in the striatum, and reduced behavioral impairment in MPTP-treated mice [76].

1.4.18.2 Family: Cupressaceae, Genus: Thuja, Species: orientalis

Thuja is a genus of coniferous trees in the cypress family. *Thuja orientalis* seed extracts modulated NO, PGE2, and IL-1 β levels and iNOS, COX-2, and IL-1 β expression in LPS-stimulated BV-2 microglial cells, and reduced infarct volume after cerebral ischemia in rats [77].

1.4.19 Order: Piperales

1.4.19.1 Family: Piperaceae, Genus: Piper, Species: nigrum

Piper nigrum (black pepper) extracts modulated AChE, CRP, NF- κ B, and MCP-1 levels in the AlCl₃-induced rat model of AD [69].

1.4.19.2 Family: Piperaceae, Genus: Piper, Species: methysticum

Piper methysticum roots possess sedative and anesthetic properties. A chemically synthesized kavalactone derivative, 2',6'-dichloro-5-methoxymethyl-5,6-dehydro-kawain, modulated iNOS induction and NO production in LPS-stimulated BV-2 microglial cells, and reduced damage after oxidative stress-induced neuronal injury [78].

1.4.20 Order: Ranunculales

1.4.20.1 Family: Ranunculaceae, Genus: Nigella, Species: sativa

Nigella sativa is an annual flowering plant in South and South West Asia. Extracts modulated a scopolamine-induced increase in MDA and oxidative stress and reduced spatial memory impairment in scopolamine-treated rats [79].

1.4.21 Order: Rosales

1.4.21.1 Family: Cannabaceae, Genus: Cannabis, Species: sativa

The cannabis plant contains molecules (e.g., 2-arachidonoyl glycerol, anandamide) that bind to G-protein-coupled cannabinoid receptors, and have been reported to reduce the progression of neurodegeneration [80].

1.4.21.2 Family: Moraceae, Genus: Morus, Species: alba

Mulberry leaves have been reported to possess antiamyloidogenic effects. Mulberry leaf and silkworm excreta extracts modulated astrocyte and microglial reaction and reduced memory impairment in A β -treated mice [81].

1.4.21.3 Family: Rosaceae, Genus: Rosa, Species: laevigata

Rosa laevigata Michx fruit flavonoids modulated DNA and mitochondrial damage, activation of Jun kinase (JNK), ERK, and p38 MAPK, and expression of cytokines after hydrogen peroxide (H₂O₂)-induced oxidative stress. Oral administration of fruit extract reduced neuronal damage following cerebral ischemia in rats [82].

1.4.21.4 Family: Urticaceae, Genus: Urtica, Species: dioica

Urtica dioica (nettle) is reported to have anti-inflammatory and antioxidant effects. It modulated ROS levels and the DNA-binding activity of NF- κ B and reduced neuronal injury after N-methyl-d-aspartate-induced excitotoxic injury in rats [83].

1.4.21.5 Family: Rosaceae, Genus: Crataegus, Species: oxyacantha

Hawthorn ethanolic extract pretreatment of 100 mg/kg for 15 days modulated proinflammatory cytokine and intercellular adhesion molecule 1 (ICAM-1) expression, and reduced the number of apoptotic cells after cerebral ischemia in rats [84].

1.4.22 Order: Sapindales

1.4.22.1 Family: Burseraceae, Genus: Commiphora, Species: erythraea

Plants in the myrrh family are reported to have anti-inflammatory effects. Extracts modulated ROS and NO levels in LPS-stimulated BV-2 microglial cells and NF- α and IL-1 β levels in the brains of LPS-treated mice [85].

1.4.22.2 Family: Anacardiaceae, Genus: Mangifera, Species: indica

Mangifera indica is a species of mango. An aqueous extract, Vimang, is traditionally used in Cuba for its anti-inflammatory and antioxidant effects. Extracts modulated TNF- α -induced inhibitor κ B (I κ B) degradation, binding of NF- κ B to DNA, and transcription of genes involved in oxidative stress [86].

1.4.22.3 Family: Anacardiaceae, Genus: Pistacia, Species: lentiscus

Pistacia lentiscus (essential oil), a mixture of terpenes and sesquiterpenes, prevented bilateral common carotid artery occlusion-induced loss of DHA. Treatment with *Pistacia lentiscus* extracts modulated an increase in COX-2 and decrease in DHA levels following cerebral ischemia in rats [87].

1.4.22.4 Family: Meliaceae, Genus: Toona, Species: sinensis

Toona sinensis leaf extract suppressed NO production, TNF- α secretion, and iNOS protein expression in LPS-stimulated microglia. Extracts also modulated nitrate, COX-1 and thromboxane levels, and reduced infarct volume after cerebral ischemia in rats [88].

1.4.23 Order: Saxifragales

1.4.23.1 Family: Crassulaceae, Genus: Graptopetalum, Species: paraguayense Graptopetalum paraguayense E. Walther is reported to have anti-inflammatory and antioxidant effects. Graptopetalum paraguayense E. Walther leaf extracts modulated ERK expression, and reduced neuronal injury after cerebral ischemia in rats [89].

1.4.23.2 Family: Paeoniaceae, Genus: Paeonia, Species: lactiflora

A component of *Paeonia lactiflora* Pall, paeoniflorin, has anti-inflammatory effects. Paeoniflorin modulated IL-1 β , TNF- α , ICAM-1, and MPO levels, and reduced neurological deficits after cerebral ischemia in rats [90].

1.4.23.3 Family: Crassulaceae, Genus: Rhodiola, Species: rosea

Extracts of *Rhodiola rosea* modulated iNOS and cytokine levels in LPS-stimulated BV-2 microglial cells and reduced iNOS, IL-1 β , and TNF- α expression in the prefrontal cortex [91].

1.4.24 Order: Solanales

1.4.24.1 Family: Solanaceae, Genus: Lycium, Species: barbarum

Lycium barbarum is one of two species of boxthorn from which wolfberry is harvested. Extracts reduced oxidative stress and protected retinal ganglion cells from secondary injury after partial optic nerve transection in rats [92].

20 Neuroprotective Effects of Phytochemicals in Neurological Disorders

1.4.24.2 Family: Solanaceae, Genus: Withania, Species: somnifera

Ashwagandha (Sanskrit for "horse smell") is referred to in Ayurveda as "Indian ginseng." It had potent anti-inflammatory action against microglial cells and may have protective effects in a model of AD (G.Y. Sun, unpublished).

1.4.24.3 Family: Convolvulaceae, Genus: Ipomoea, Species: batatas

Purple sweet potato color (PSPC), a naturally occurring anthocyanin, modulated glial fibrillary acidic protein (GFAP) expression, iNOS, and COX-2 levels and improved performance in the open field and passive avoidance tests in D-galactose-treated mice [93].

1.4.25 Order: Vitales

1.4.25.1 Family: Vitaceae, Genus: Vitis, Species: vinifera

Red grapes contain resveratrol, which prevented nuclear translocation of NF- κ B by inhibiting I κ B kinase via activation of an enzyme of the NAD⁺-dependent histone deacetylase, SIRT1; this enzyme deactivates NF- κ B via deacetylation, leading to modulation of gene-expression changes associated with aging in mice [94].

1.4.26 Order: Zingiberales

1.4.26.1 Family: Zingiberaceae, Genus: Curcuma, Species: longa

Curcuma longa ("yellow ginger" in Chinese or "kunyit" in Malay) is reported to have anti-inflammatory and antioxidant effects. Curcumin modulated TNF-dependent activation of NF- κ B through inhibition of p65 translocation to the nucleus and I κ B- α degradation [95]. It also modulated tau phosphorylation and A β formation [96].

1.4.26.2 Family: Zingiberaceae, Genus: Zingiber, Species: officinale

Gingerols and shogaols in ginger modulated NF- κ B, IL-1 β , and TNF- α expression in microglial cells and human monocytic THP-1 cells, and reduced neuronal damage after 1-methyl-4-phenylpyridinium (MPP⁺) treatment [97].

1.4.27 Order: Fucales

1.4.27.1 Family: Sargassaceae, Genus: *Myagropsis*, Species: *myagroides Myagropsis myagroides* brown algae ethanolic extract modulated iNOS and COX-2 mRNA and protein expression in LPS-stimulated BV-2 microglial cells [98].

1.4.28 Order: Agaricales

1.4.28.1 Family: Agaricaceae, Genus: Agaricus, Species: bisporus

Agaricus bisporus (button mushroom, portobello mushroom) is abundant in ergosterol, which can be converted to vitamin D2 under ultraviolet (UV) light. Treatment with vitamin D2-enriched mushrooms resulted in improved learning and memory, in a mouse model of AD [99].

1.4.29 Order: Polyporales

1.4.29.1 Family: Ganodermataceae, Genus: Ganoderma, Species: lucidum

Ganoderma lucidum mushroom extracts modulated NO, TNF- α , and IL-1 β levels in LPS-stimulated microglial cells, suggesting that *Ganoderma lucidum* is a promising agent for the treatment of neuroinflammation [100].

1.5 Use of Phytochemicals against Neuroinflammation

1.5.1 Catechin Flavonoid Polyphenols

The beneficial effects of green tea have been attributed to the interactions of green-tea catechins with cellular proteins. These interactions lead to changes in enzyme activity and ligand/receptor function. EGCG is converted to a catechol-quinone upon autooxidation, and the resultant quinone moiety rapidly reacts with the sulfhydryl group of proteins to form cysteinyl-flavonoid adducts [101]. In addition, EGCG binds to serum proteins such as fibronectin, fibrinogen, histidine-rich glycoproteins, 67-kDa laminin receptor, Bcl-2 proteins, and vimentin. EGCG interacts with growth factor receptors (e.g., epidermal growth factor, platelet-derived growth factor (PDGF), IGF-1, and vascular endothelial growth factor (VEGF) receptors) and alters signal transduction processes. The ability of EGCG to cross the BBB allows its use as a preventive treatment for neurodegenerative diseases. EGCG may work through a voltage-gated sodium channel signaling pathway. It also inhibits the activity of HSP90 by directly binding to HSP70-interacting protein. Green-tea components inhibit the arachidonic acid pathway (PLA2, COX, and LOX) and decrease the production of prostaglandins and leukotrienes, key mediators of the acute inflammatory cascade. Together, results suggest that green-tea components inhibit inflammation by downregulating the expression of proinflammatory enzymes and cytokines [102].

Catechins in green tea have similar neuroprotective effects, and their antioxidantand free radical-scavenging properties are well known [103,104]. Catechins function as metal chelators that quench copper (II) and iron (III) ions to form inactive complexes and prevent the generation of toxic free radicals. In addition, ultra-rapid electron transfer from catechins to ROS-induced radical sites on DNA can occur. The anti-inflammatory effects of catechins may involve downregulation of NO synthase activity and scavenging of NO. This can occur via attenuation of signal transducer and activator of transcription (STAT)-1 α activation or through prevention of I κ B degradation which inhibits NF- κ B from binding to the promoter region of the NO synthase gene [105].

1.5.2 Anthocyanin Flavonoid Polyphenols

Fruits such as blueberries contain anthocyanin flavonoid polyphenols, which offer beneficial effects for memory [106,107]. Anthocyanins suppress apoptosis resulting from mitochondrial oxidative stress, while anthocyanin and pro-anthocyanin-rich extracts prevent death of dopaminergic neurons caused by rotenone, via an improvement in mitochondrial function. Blueberry extracts enhance microglial clearance of A β -inhibited aggregation of A β 1–42 and suppress microglial activation via an effect on p44/42 MAPK signaling. Extracts also modulate inflammatory cytokine IL-1 β and TNF- α expression, augment expression of the neurotrophic factor IGF-1 in the hippocampus, and improve cognitive performance in rats after excitotoxic injury induced by kainate [51].

1.5.3 Stilbenoid Polyphenols

The beneficial effects of resveratrol result from its antiaging, anticarcinogenic, cardioprotective, and neuroprotective activities, which are supported by its anti-inflammatory, antioxidant, and gene-modulating properties [108–110]. Due to its structural similarity

22 Neuroprotective Effects of Phytochemicals in Neurological Disorders

to diethylstilbestrol (a synthetic estrogen), resveratrol produces estrogenic effects by binding to estrogen receptors and evoking neurochemical effects that parallel those exerted endogenously. Many of these effects are associated with its ability to inhibit transcription factors such as MAPK, AP-1, and NF-κB. Resveratrol prevents nuclear translocation of NF- κ B by inhibiting I κ B kinase. This mechanism is postulated to occur via activation of an NAD + -dependent histone deacetylase, SIRT1, which deactivates NF-KB via deacetylation [94]. Resveratrol also induces antioxidant enzymes such as catalase, SOD, and GPx, as well as HO-1. The anti-inflammatory effects of resveratrol are mediated through downregulation of TNF- α , COX-2, iNOS, IFN- γ , and various interleukins. Resveratrol (5 mg/kg for males, 1 mg/kg for females) modulates IL-1 β and TNF- α level, microglial activation, and ROS production in the ischemic cortex and reduces infarct volumes after ischemic stroke in mice. In addition, resveratrol (0.01% by weight in the diet for 1 year) reduces average A β plaque density and modulates the microglial activation in the transgenic amyloid beta protein protein/presenilin-1 (A β PP/PS1) mouse model of AD [111]. It also prevents the proinflammatory effect of fibrillary A β on macrophages by potently inhibiting the effect of A β on I κ B phosphorylation, the activation of STAT1 and STAT3, and the secretion of TNF- α and IL-6 secretion [112].

1.5.4 Curcuminoid Polyphenols

The yellow curcumin pigment, bis-(4-hydroxy-3-methoxyphenyl)-1,6-diene-3,5-dione, in turmeric (Curcuma longa) is reported to have anti-inflammatory and antioxidant properties. Curcumin inhibits TNF-dependent activation of NF-κB by inhibiting p65 translocation to the nucleus and the degradation of $I\kappa B-\alpha$ [95]. This may occur via quenching of reactive oxygen intermediates. In addition, curcumin blocks the DNA binding of JNK/AP-1 transcription factor and downregulates c-Jun by blocking its transcription. It also acts as a peroxisome proliferator-activated receptor gamma (PPARy) agonist, inhibiting activation and inflammation of NF- κ B. Curcumin reduces the levels of cytokines (e.g., IL-1 β , TNF- α) and other inflammatory factors (e.g., iNOS) and inhibits various factors of the inflammatory pathway (e.g., COX-2 and LOX) at the transcriptional level. It activates Nrf2, leading to increased expression of HO-1, an enzyme that plays a pivotal role in cytoprotection against noxious stimuli. Curcumin modulates NF-kB activation and subsequent ICAM-1 gene expression in cultured brain microvessel endothelial cells. It also reduces neutrophil adhesion to the cerebrovascular endothelium and TNF- α and ICAM-1 expression in the brain after experimental stroke induced by middle cerebral artery occlusion in rats. It modulates Aβ-stimulated inflammatory responses in primary astrocytes and reduces GFAP expression and improved spatial memory in the $A\beta 1-40$ -induced rat model of AD. It also modulates COX-2 and GFAP in A β 25–35-treated astrocytes and the antineuroinflammatory effects of a PPARy antagonist, GW9662 [113].

1.5.5 Ginkgo biloba Polyphenols

The active compound of GB, EGb 761, possesses potent antioxidant, memory-enhancing, anti-inflammatory, and blood flow-promoting properties, which play important roles in modulating brain activities such as cognition, concentration, mental alertness, and mental fatigue. Many of these activities are mediated by interactions between constituents of EGb 761 (bilobalide) and GABA and glycine receptors located on neuronal cell membranes. These receptors play an important role in memory formation, consolidation, and cognition [114,115]. EGb 761 also enhances cholinergic processes in various cortical regions. Together, results support the view that the psychological and physiological benefits of EGb 761 are partly due to modulation of neurotransmitters and neurotransmitter receptors. EGb 761 also benefits the microcirculation by improving blood flow in small vessels. It exerts its antioxidant and anti-inflammatory effects via activation of the HO-1/Nrf2 pathway, VEGF regulation, and downregulation of various inflammatory mediators. Its antioxidative action is suggested to work in concert with its antiapoptotic mechanism. The anti-inflammatory effects of the GB polysaccharide are shown by its suppression of NO production. Bilobalide (4, 8 mg/kg) extracted from GB leaves modulates TNF- α and A β 1–40 expression, reduces neuronal damage in the frontal cortex and hippocampus, and protects against learning and memory impairments in a rat model of AD [116].

1.5.6 Aromatic Acid Class of Phenolic Compounds

Cinnamon (Figure 1.4) produces anti-inflammatory, antimicrobial, antioxidant, antitumor, cardiovascular, cholesterol-lowering, and immunomodulatory effects. *In vitro* studies have demonstrated that it may act as an insulin mimetic to potentiate insulin activity or stimulate cellular glucose metabolism [117,118]. Cinnamon not only scavenges ROS and NO, but also interacts with superoxide anion and peroxynitrite. Cinnamaldehyde is an anti-inflammatory constituent in cinnamon and together with 2-methoxycinnamaldehyde, are potent NF- κ B inhibitors. They suppresses TLR4 oligomerization and attenuates LPS-induced intracellular signaling processes in peripheral macrophages. Cinnamaldehyde also reduces LPS-induced intracellular ROS formation, thereby attenuating oxidative stress-triggered signal transduction pathways such as the NF- κ B-inducing kinase/I κ B- α kinase, ERK, and p38 MAPK pathways. Cinnamic acid and its derivatives significantly (~12–63%) inhibit the formation of advanced glycation end products, in a concentration-dependent manner [119].

1.5.7 Phenylethanoid Class of Phenolic Compounds

Olive oil contains phenolic compounds that are well-known antioxidants. Extra-virgin olive oil fed to transgenic mice with memory impairments improves memory. This is associated with a reduction in oxidative stress, and increases in brain glutathione and GR levels [120]. Post-ischemic neuronal injury may be ameliorated by olive oil via the combination of its effects on the cholinergic system and its antioxidant effects. Long-term consumption of olive oil increases the proportion of monounsaturated fatty acids, particularly oleic acid, while reducing the level of arachidonic acid in the cell membrane, suggesting its potential to modulate the production of proinflammatory eicosanoids. Tyrosol and hydroxytyrosol isolated from olive oil decrease the nuclear translocation of the NF- κ B subunits following A β exposure *in vitro*, suggesting the involvement of NF- κ B in neuroprotective effects of olive oil [121].

1.5.8 Organosulfur Class of Glucosinolates

The beneficial effects of garlic in human health are due to its anti-inflammatory, antioxidant, anticancer, antifungal, and immune system-enhancing properties, as well as its inhibition of prostaglandin production. Organosulfur compounds in garlic target



Figure 1.4 Chemical structures of components of garlic and cinnamon.

multiple signal transduction pathways and regulate the expression of many genes and the induction of many enzymes (e.g., arylamine N-acetyltransferase, SOD-like activity, H_2O_2 -scavenging activity, GSH redox cycle enzymes, cytochrome P450 reductase, and lactate dehydrogenase) in the brain, liver, and other visceral tissues [122,123]. SAC (Figure 1.4) possesses neuroprotective properties, including antioxidant and radical scavenging effects [124]. SAC and AGE inhibit apoptosis by preventing caspase-3 activation. Garlic compounds modulate intracellular levels of GSH. Low levels of GSH are present in the Alzheimer's brain and are elevated by AGE. The antioxidant action of AGE is demonstrated by the preservation of expression of GPx and GR. This is partly caused by inhibition of NF- κ B via interference with an intermediate of TNF- α . An alternative explanation for the antioxidant activity is that SAC inhibits NF- κ B activation induced by TNF- α and H₂O₂. A sulfur compound isolated from garlic, thiacremonone, modulates phosphorylation of IkBa and NF-kB activation and reduces LPS-induced amyloidogenesis in cultured astrocytes and BV-2 cells. Results suggest the potential of thiacremonone for intervention in neuroinflammatory diseases, including AD [125].

1.5.9 n-3 Fatty Acids

n-3 fatty acids such as DHA are abundant in fish such as salmon, tuna, and whitefish. In plant products such as walnuts and flaxseed oil, the main n-3 fatty acid is alphalinolenic acid (ALA), which is converted in the body to EPA and DHA [126]. The presence of ALA is crucial during periods of active growth around birth. Its conversion to EPA and DHA occurs in many body tissues, including the liver and brain – but at a restricted rate. The use of ALA labeled with radioisotopes suggests that with a background diet high in saturated fat, the conversion of ALA to long-chain metabolites is approximately 6% for EPA and 3.8% for DHA. This low rate of ALA to DHA conversion is due not only to a high percentage of ALA being directed toward β -oxidation, but also lower activities of enzymes that convert ALA to DHA in humans compared to rats. n-3 fatty acids are metabolized to lipid mediators, such as resolvins and neuroprotectins. These lipid mediators not only regulate inflammatory cytokines and chemokines (TNF- α , IL-1 β , IL-10) but also block apoptotic cell death caused by inflammation and oxidative stress. DHA inhibits arachidonic acid metabolism and downregulates the expression of COX-2 via NF- κ B. Mice fed with fish oil showed decreased production of TNF, IL-1 β , and IL-6 by endotoxin-stimulated macrophages, and reduced serum levels of TNF, IL-1 β , and IL-6 were found in mice injected with endotoxin. DHA is converted to neuroprotectin D1, which confers protection from apoptosis induced by oxidative stress and anti-inflammatory pathways. DHA also reduces oxidative stress by decreasing ROS production by mitochondria. A 3-month DHA supplementation significantly altered the n-3:n-6 polyunsaturated fatty acid ratio in the brain, increased levels of the antiapoptotic molecule Bcl-2 in the brain, modulated COX2 and IL-1 β levels and microglial activation after ischemic injury, and decreased infarct volume following middle cerebral artery occlusion in rats. Results suggest that diet-induced accumulation of DHA in the brain may protect against immune response/brain damage in ischemic stroke. DHA and curcumin supplementation also improved cognitive function and modulates $A\beta$ accumulation, oxidative damage, and synaptic deficits in mouse models of AD [127].

1.6 Phytochemicals and Stroke

1.6.1 Tea

Many preclinical studies have found that tea components are effective in reducing stroke volume following middle cerebral artery occlusion, and decrease in infarct volume are found with both tea extracts consumed orally and tea components introduced intraperitoneally in rodent models such as rats, mice, and gerbils. Epidemiological studies support this finding in humans, and are consistent across countries and types of tea [128]. Current evidence indicates beneficial effects of tea and cocoa on endothelial function and total and low-density lipoprotein (LDL) cholesterol [129]. Higher green tea and coffee consumption is inversely associated with risk of cardiovascular disease and stroke in the general population [130]. In addition, daily consumption of four or more cups of black tea is inversely associated with risk of stroke [131]. A significant decrease in ischemic stroke risk was observed for drinking at least one cup of tea weekly when compared with infrequent or nondrinkers, the risk reduction being largest when drinking one to two cups of green or oolong (black) tea daily [132,133].

1.6.2 Flavonoids

Intake of polyphenols, especially from lignans, flavanols, and hydroxybenzoic acids, is associated with decreased cardiovascular disease risk [134]. High intake of flavonoids was associated with decreased risk of ischemic stroke and possibly with reduced cardiovascular disease mortality in one study [135], although another found that total flavonoid intake was not inversely associated with risk of stroke but that increased intake of citrus fruits/juices (the main dietary source of flavanones) was correlated with reduced ischemic stroke risk [136].

1.6.3 Resveratrol

Resveratrol has shown potential for treatment of stroke in animal and *in vitro* human cell studies [137]. Resveratrol improved memory performance in association with improved glucose metabolism and increased hippocampal functional connectivity in older adults [138].

1.6.4 Ginkgo biloba

Studies on the effect of GB on functional outcome in patients with acute stroke suggest that extracts may have protective effects in ischemic stroke [139]. In contrast, another study reported no convincing evidence to support the use of GB for promotion of stroke recovery [140].

1.6.5 Olive Oil

Studies support an inverse association of olive oil consumption with stroke, but not with coronary heart disease [141]. High olive oil consumption and high plasma oleic acid (as an indicator of olive oil intake) were associated with lower incidence of stroke in older adults [142]. One study reported that among persons at high cardiovascular risk, a Mediterranean diet supplemented with extra-virgin olive oil or nuts reduced the incidence of major cardiovascular events [143].

1.6.6 n-3 Fatty Acids

Dietary supplementation with n-3 fatty acids did not reduce the risk of cardiovascular disease in elderly participants with age-related macular degeneration [144]. Similarly, there was no difference between n-3 fatty acids and placebo in end points such as mortality, nonfatal stroke, nonfatal acute myocardial infarction, systemic embolism, heart failure development, or recurrent atrial fibrillation [145,146]. In a large general-practice cohort of patients with multiple cardiovascular risk factors, daily treatment with n-3 fatty acids did not reduce cardiovascular mortality or morbidity, including nonfatal myocardial infarction and nonfatal stroke [147]. Meta-analyses showed no overall association between n-3 fatty acid intake and stroke [148], and no effect on cardiovascular biomarkers or mood was found following treatment of post-ischemic stroke patients with moderate-dose fish-oil supplements [149].

1.7 Phytochemicals and AD

1.7.1 Flavonoids

Intake of flavonols and of combined flavonoids (all five combined) were the two parameters among dietary factors that were inversely correlated with dementia in studies among 23 developed countries. Results suggest that higher consumption of dietary flavonoids (especially flavonols) is associated with lower population rates of dementia [150]. One study reported that regular cocoa flavonol consumption could reduce agerelated cognitive dysfunction, possibly through an improvement in insulin sensitivity [151]. Flavonol consumption results in enhanced dentate gyrus (DG) function, as shown by functional magnetic resonance imaging (fMRI) and cognitive tests. Together, results suggest that dietary flavonols may be beneficial in modulating age-related cognitive deficits, through an effect on the DG [152].

1.7.2 Resveratrol

Recent epidemiological evidence has revealed the protective role of dietary polyphenols from grape products against AD-type cognitive deterioration, which stems in part from interference with the generation and assembly of A β peptides into neurotoxic oligomeric aggregated species. *In vivo* data have demonstrated the neuroprotective properties of resveratrol in animal models of stress and disease [11]. Resveratrol promotes nonamyloidogenic cleavage of amyloid precursor protein and clearance of A β [153], and recent studies have shown a role for grape-derived preparations in reducing tau aggregation, which may be useful in the prevention and treatment of AD [154].

1.7.3 Curcumin

Curcumin has antiamyloidogenic, anti-inflammatory, antioxidative, and metal-chelating properties that may result in potential neuroprotective effects. However, it exhibits very low bioavailability, mainly due to its poor aqueous solubility, poor stability in solution, and rapid intestinal first-pass effect and hepatic metabolism [155]. At present, four clinical trials concerning the effects of curcumin on AD have been conducted. Two of them (performed in China and the United States) report no significant differences in changes in cognitive function between curcumin and placebo, while results of the other two are not yet available. Additional trials are necessary to determine the potential usefulness of curcumin for the prevention and treatment of AD [156].

1.7.4 Ginkgo biloba

GB is one of the most investigated and adopted herbal remedies for AD. A 24-week randomized controlled trial was conducted to assess the efficacy of a 240 mg once-daily preparation of GB extract EGb 761 in 404 outpatients aged \geq 50 years, diagnosed with mild to moderate dementia, AD, or vascular dementia with neuropsychiatric features [157]. Treatment with EGb 761 at a once-daily dose of 240 mg was safe and resulted in improvement in cognition, psychopathology, functional measures, and quality of life among patients and caregivers [158]. Significant changes in Mini-Mental State Examination (MMSE) score over a 12-month follow-up period were reported between patients on combined therapy and those taking only cholinesterase inhibitors,

28 Neuroprotective Effects of Phytochemicals in Neurological Disorders

suggesting that GB may provide some added cognitive benefits in AD patients already under cholinesterase inhibitor treatment [159]. EGb 761 240 mg once-daily was superior to placebo in treatment of patients with dementia with neuropsychiatric symptoms [160], and improvement in quality of life and cognitive function were noted with GB in irradiated brain-tumor patients [161]. A randomized, double-blind exploratory trial was undertaken to compare the treatment effects and tolerability of EGb 761, donepezil, and combined treatment in patients with AD and neuropsychiatric features. This study suggested no difference in the efficiency of EGb 761 and donepezil, and that combination therapy might be better than monotherapy due to having fewer side effects [162]. Another study directly compared a cholinesterase inhibitor with GB, and found no difference in efficacy of EGb 761 and donepezil for treatment of mild to moderate Alzheimer's dementia [163]. A meta-analysis looking at the prevention effect of ginkgo against AD suggests that GB may help established AD patients with cognitive symptoms but cannot prevent the neurodegenerative progression of the disease [164]. A trial of more than 2000 participants, of whom 1406 received at least one dose of GB extract and 1414 received at least one dose of placebo 2000 reported that long-term use of standardized GB extract did not reduce the risk of progression to AD [165].

1.7.5 n-3 Fatty Acids

Oral supplementation with n-3 fatty acids conferred changes in the n-3 fatty acid profile in the cerebrospinal fluid (CSF), suggesting transfer of these fatty acids across the BBB [166]. The effects of supplementation with n-3 fatty acids alone or with n-3 fatty acids plus α -lipoic acid were compared to placebo by MMSE, Activities of Daily Living/ Instrumental Activities of Daily Living (ADL/IADL), and Alzheimer Disease Assessment Scale – cognitive subscale (ADAS-cog). Results indicate that the n-3 plus lipoic acid group showed fewer declines in MMSE and IADL, and that the n-3 group had lower decline in IADL compared to placebo controls [167]. n-3 fatty acid supplementation for 6 months increased plasma levels of transthyretin in patients with AD. Transthyretin binds to A β and may influence A β deposition in the brain [168]. n-3 fatty acid supplementation also resulted in significant increases in DHA and EPA plasma concentrations and modulation of genes involved in inflammation and neurodegeneration (e.g., CD63, MAN2A1, CASP4, LOC399491, NAIP, and SORL1). Results suggest that dietary n-3 fatty acid supplementation affects the expression of inflammatory-related genes that might have an impact on AD [169]. The erythrocyte membranes of subjects on a DHAphospholipids-, melatonin-, and tryptophan-supplemented diet showed significant increases in eicosapentenoic acid, docosapentenoic acid, and DHA concentrations, but decreases in arachidonic acid, MDA, and lipofuscin levels [170]. DHA also inhibited inflammatory cytokines in cells from subjects with AD [171]. One study reported the potential role of fish oil in improving memory function in mild cognitive impairment (MCI) subjects [172], while another reported that 24-week supplementation with 900 mg/d DHA improved learning and memory function in age-related cognitive decline [173]. Other studies, however, reported that supplementation with DHA did not slow the rate of cognitive and functional decline [174] or that supplementation with n-3 fatty acid did not result in marked effects on neuropsychiatric symptoms in patients with mild to moderate AD [175]. An intervention study reported that only ApoE4 noncarriers had increased concentrations of long-chain n-3 fatty acids in response to supplementation. The mechanisms underlying this gene-by-diet interaction may involve impaired fatty acids and cholesterol transport or altered metabolism of n-3 fatty acids [176]. Studies using Souvenaid, a combination of uridine monophosphate, choline, EPA, DHA, phospholipids, vitamin C, vitamin E, selenium, vitamin B6, vitamin B12, and folic acid showed preservation of the organization of brain networks in patients with mild AD within 24 weeks, suggesting that this combination may be useful for modulating disease progression in AD [177].

1.8 Conclusion

Plants and phytochemicals not only provide beneficial effects in normal aging, but also modulate or delay the onset of neurodegenerative diseases. Phytochemicals produce actions on a wide spectrum of molecular targets. Many of these inhibit oxidative stress by scavenging free radicals and neuroinflammation, and by stimulating antiinflammatory responses. They act via downregulation of proinflammatory enzymes through activation of PPAR γ ; inhibition of PI3K, tyrosine kinases, NF- κ B, and c-Jun; modulation of cell survival/cell-cycle genes; and stimulation of ARE pathways. Phytochemicals disrupt the Nrf2–Keap1 association, thereby releasing Nrf2, which translocates to the nucleus and upregulates the expression of phase II detoxifying enzymes, such as HO-1 and GSTs, which have a protective effects on cells. Clinical evidence suggests the use of certain phytochemicals for the prevention of stroke and/ or AD. Further work needs to be carried out to comprehensively evaluate each phytochemical in terms of its dose–response, bioavailability, safety, and effectiveness with regard to its anti-inflammatory and antioxidative properties, as well as its disease-prevention and therapeutic effects.

Acknowledgments

This work was supported by grants from the National Medical Research Council and the National University Health System, Singapore.

Conflicts of Interest

The authors have no conflicts of interest.

References

- 1 Chakraborty S, Kaushik DK, Gupta M, Basu A. Inflammasome signaling at the heart of central nervous system pathology. *J Neurosci Res* 2010; 88: 1615–1631.
- 2 Farooqui AA. *Neurochemical Aspects of Neurotraumatic and Neurodegenerative Diseases*, Springer, New York, 2010.
- **3** Farooqui AA. *Phytochemicals, Signal Transduction, and Neurological Disorders,* Springer, New York, 2012.

- **4** Virgili F, Marino M. Regulation of cellular signals from nutritional molecules: a specific role for phytochemicals, beyond antioxidant activity. *Free Radic Biol Med* 2008; 45: 1205–1216.
- **5** Spencer JP. Beyond antioxidants: the cellular and molecular interactions of flavonoids and how these underpin their actions on the brain. *Proc Nutr Soc* 2010; 69: 244–260.
- **6** Spencer JP, Vauzour D, Rendeiro C. Flavonoids and cognition: the molecular mechanisms underlying their behavioural effects. *Arch Biochem Biophys* 2009; 492: 1–9.
- 7 Wakabayashi N, Slocum SL, Skoko JJ, Shin S, Kensler TW. When NRF2 talks, who's listening? *Antioxid Redox Signal* 2010; 13: 1649–1663.
- 8 Mattson MP. Hormesis defined. Ageing Res Rev 2008; 7: 1–7.
- **9** Mattson MP, Cheng A. Neurohormetic phytochemicals: low-dose toxins that induce adaptive neuronal stress responses. *Trends Neurosci* 2006; 29: 632–639.
- 10 Mattson MP, Maudsley S, Martin B. BDNF and 5-HT: a dynamic duo in age-related neuronal plasticity and neurodegenerative disorders. *Trends Neurosci* 2004; 27: 589–594.
- 11 Vingtdeux V, Dreses-Werringloer U, Zhao H, Davies P, Marambaud P. Therapeutic potential of resveratrol in Alzheimer's disease. *BMC Neurosci* 2008; 9 (Suppl. 2): S6.
- 12 Ghalandarlaki N, Alizadeh AM, Ashkani-Esfahani S. Nanotechnology-applied curcumin for different diseases therapy. *BioMed Research International* 2014; 2014: 23.
- 13 Lee WH, Loo CY, Young PM, Traini D, Mason RS, Rohanizadeh R. Recent advances in curcumin nanoformulation for cancer therapy. *Expert Opin Drug Deliv* 2014; 11: 1183–1201.
- 14 Wang Z, Chen C, Zhang Q, Gao M, Zhang J, Kong D, Zhao Y. Tuning the architecture of polymeric conjugate to mediate intracellular delivery of pleiotropic curcumin. *Eur J Pharm Biopharm* 2015; 90: 53–62.
- 15 Feng WY. Metabolism of green tea catechins: an overview. *Curr Drug Metab* 2006; 7: 755–809.
- 16 Leonarduzzi G, Testa G, Sottero B, Gamba P, Poli G. Design and development of nanovehicle-based delivery systems for preventive or therapeutic supplementation with flavonoids. *Curr Med Chem* 2010; 17: 74–95.
- 17 Viskupicova J, Ondrejovic M, Sturdik E. The potential and practical applications of acylated flavonoids. *Pharmazie* 2009; 64: 355–360.
- 18 Smith A, Giunta B, Bickford PC, Fountain M, Tan J, Shytle RD. Nanolipidic particles improve the bioavailability and alpha-secretase inducing ability of epigallocatechin-3-gallate (EGCG) for the treatment of Alzheimer's disease. *Int J Pharm* 2010; 389: 207–212.
- 19 Walle T, Hsieh F, DeLegge MH, Oatis JE Jr., Walle UK. High absorption but very low bioavailability of oral resveratrol in humans. *Drug Metab Dispos* 2004; 32: 1377–1382.
- **20** Wenzel E, Somoza V. Metabolism and bioavailability of trans-resveratrol. *Mol Nutr Food Res* 2005; 49: 472–481.
- **21** Juan ME, Maijo M, Planas JM. Quantification of trans-resveratrol and its metabolites in rat plasma and tissues by HPLC. *J Pharm Biomed Anal* 2010; 51: 391–398.
- 22 Cheng AL, Hsu CH, Lin JK, Hsu MM, Ho YF, Shen TS, Ko JY, Lin JT, Lin BR, Ming-Shiang W, Yu HS, Jee SH, Chen GS, Chen TM, Chen CA, Lai MK, Pu YS, Pan MH, Wang YJ, Tsai CC, Hsieh CY. Phase I clinical trial of curcumin, a chemopreventive agent, in patients with high-risk or pre-malignant lesions. *Anticancer Res* 2001; 21: 2895–2900.

- **23** Sharma RA, McLelland HR, Hill KA, Ireson CR, Euden SA, Manson MM, Pirmohamed M, Marnett LJ, Gescher AJ, Steward WP. Pharmacodynamic and pharmacokinetic study of oral Curcuma extract in patients with colorectal cancer. *Clin Cancer Res* 2001; 7: 1894–1900.
- **24** Ray B, Bisht S, Maitra A, Maitra A, Lahiri DK. Neuroprotective and neurorescue effects of a novel polymeric nanoparticle formulation of curcumin (NanoCurc) in the neuronal cell culture and animal model: implications for Alzheimer's disease. *J Alzheimers Dis* 2011; 23: 61–77.
- **25** Goh LML, Barlow PJ. Flavonoid recovery and stability from Ginkgo biloba subjected to a simulated digestion process. *Food Chemistry* 2004; 86: 195–202.
- **26** Biber A. Pharmacokinetics of Ginkgo biloba extracts. *Pharmacopsychiatry* 2003; 36(Suppl. 1): S32–S37.
- 27 Woelkart K, Feizlmayr E, Dittrich P, Beubler E, Pinl F, Suter A, Bauer R. Pharmacokinetics of bilobalide, ginkgolide A and B after administration of three different Ginkgo biloba L. preparations in humans. *Phytother Res* 2010; 24: 445–450.
- **28** Lang D, Ude C, Wurglics M, Schubert-Zsilavecz M, Klein J. Brain permeability of bilobalide as probed by microdialysis before and after middle cerebral artery occlusion in mice. *J Pharm Pharm Sci* 2010; 13: 607–614.
- **29** Loew D. [Value of Ginkgo biloba in treatment of Alzheimer dementia.] *Wien Med Wochenschr* 2002; 152: 418–422.
- **30** Nagae S, Ushijima M, Hatono S, Imai J, Kasuga S, Matsuura H, Itakura Y, Higashi Y. Pharmacokinetics of the garlic compound S-allylcysteine. *Planta Med* 1994; 60: 214–217.
- **31** Freeman F, Kodera Y. Garlic chemistry: stability of S-(2-propenyl)-2-propene-1-sulfinothioate (allicin) in blood, solvents, and simulated physiological fluids. *J Agric Food Chem* 1995; 43: 2332–2338.
- 32 Agapakis C. Allegery recapitulates phylogeny. *Scientific American* 2011; Sep. 18. Available from: http://blogs.scientificamerican.com/oscillator/allergy-recapitulatesphylogeny/(last accessed August 26, 2016).
- **33** Berger KJ, Guss DA. Mycotoxins revisited: Part I. *J Emerg Med* 2005; 28: 53–62.
- 34 Gallego Dominguez S, Suarez Santisteban MA, Luengo Alvarez J, Gonzalez Castillo P, Castellano Cervino I. [Acute renal failure after intake of mushrooms: the orellanus syndrome.] *Nefrologia* 2008; 28: 351–352.
- **35** Li W, Chu Y, Zhang L, Yin L, Li L. Ginsenoside Rg1 prevents SK-N-SH neuroblastoma cell apoptosis induced by supernatant from Abeta1-40-stimulated THP-1 monocytes. *Brain Res Bull* 2012; 88: 501–506.
- 36 Bu Y, Jin ZH, Park SY, Baek S, Rho S, Ha N, Park SK, Kim H. Siberian ginseng reduces infarct volume in transient focal cerebral ischaemia in Sprague-Dawley rats. *Phytother Res* 2005; 19: 167–169.
- **37** Ramesh BN, Girish TK, Raghavendra RH, Naidu KA, Rao UJ, Rao KS. Comparative study on anti-oxidant and anti-inflammatory activities of Caesalpinia crista and Centella asiatica leaf extracts. *J Pharm Bioallied Sci* 2014; 6: 86–91.
- 38 Moon YJ, Lee JY, Oh MS, Pak YK, Park KS, Oh TH, Yune TY. Inhibition of inflammation and oxidative stress by Angelica dahuricae radix extract decreases apoptotic cell death and improves functional recovery after spinal cord injury. *J Neurosci Res* 2012; 90: 243–256.

- 32 Neuroprotective Effects of Phytochemicals in Neurological Disorders
 - 39 Poulose SM, Fisher DR, Larson J, Bielinski DF, Rimando AM, Carey AN, Schauss AG, Shukitt-Hale B. Anthocyanin-rich acai (Euterpe oleracea Mart.) fruit pulp fractions attenuate inflammatory stress signaling in mouse brain BV-2 microglial cells. *J Agric Food Chem* 2012; 60: 1084–1093.
 - 40 Leow SS, Sekaran SD, Tan Y, Sundram K, Sambanthamurthi R. Oil palm phenolics confer neuroprotective effects involving cognitive and motor functions in mice. *Nutr Neurosci* 2013; 16: 207–217.
 - **41** Ray B, Chauhan NB, Lahiri DK. The "aged garlic extract:" (AGE) and one of its active ingredients S-allyl-L-cysteine (SAC) as potential preventive and therapeutic agents for Alzheimer's disease (AD). *Curr Med Chem* 2011; 18: 3306–3313.
 - **42** Liu XL, Luo L, Liu BB, Li J, Geng D, Liu Q, Yi LT. Ethanol extracts from Hemerocallis citrina attenuate the upregulation of proinflammatory cytokines and indoleamine 2,3-dioxygenase in rats. *J Ethnopharmacol* 2014; 153: 484–490.
 - 43 Borah A, Paul R, Choudhury S, Choudhury A, Bhuyan B, Das Talukdar A, Dutta Choudhury M, Mohanakumar KP. Neuroprotective potential of silymarin against CNS disorders: insight into the pathways and molecular mechanisms of action. *CNS Neurosci Ther* 2013; 19: 847–853.
 - 44 Zeng KW, Wang S, Dong X, Jiang Y, Tu PF. Sesquiterpene dimer (DSF-52) from Artemisia argyi inhibits microglia-mediated neuroinflammation via suppression of NF-kappaB, JNK/p38 MAPKs and Jak2/Stat3 signaling pathways. *Phytomedicine* 2014; 21: 298–306.
 - **45** Elmann A, Mordechay S, Erlank H, Telerman A, Rindner M, Ofir R. Antineuroinflammatory effects of the extract of Achillea fragrantissima. *BMC Complement Altern Med* 2011; 11: 98.
 - **46** Cho J, Kim HM, Ryu JH, Jeong YS, Lee YS, Jin C. Neuroprotective and antioxidant effects of the ethyl acetate fraction prepared from Tussilago farfara L. *Biol Pharm Bull* 2005; 28: 455–460.
 - 47 Li FQ, Lu XZ, Liang XB, Zhou HF, Xue B, Liu XY, Niu DB, Han JS, Wang XM. Triptolide, a Chinese herbal extract, protects dopaminergic neurons from inflammation-mediated damage through inhibition of microglial activation. *J Neuroimmunol* 2004; 148: 24–31.
 - **48** Nerurkar PV, Johns LM, Buesa LM, Kipyakwai G, Volper E, Sato R, Shah P, Feher D, Williams PG, Nerurkar VR. Momordica charantia (bitter melon) attenuates high-fat diet-associated oxidative stress and neuroinflammation. *J Neuroinflammation* 2011; 8: 64.
 - **49** Zhang ZL, Zuo YM, Wang QH, Xiao HB, Kuang HX. [Effects of Valeriana amurensis on the expressions of iNOS, COX-2 and IkappaCB-alpha in Alzheimer's disease model rat's brain.] *Zhong Yao Cai* 2010; 33: 581–583.
 - 50 Yu Y, Wu Y, Szabo A, Wu Z, Wang H, Li D, Huang XF. Teasaponin reduces inflammation and central leptin resistance in diet-induced obese male mice. *Endocrinology* 2013; 154: 3130–3140.
 - 51 Shukitt-Hale B, Lau FC, Carey AN, Galli RL, Spangler EL, Ingram DK, Joseph JA. Blueberry polyphenols attenuate kainic acid-induced decrements in cognition and alter inflammatory gene expression in rat hippocampus. *Nutr Neurosci* 2008; 11: 172–182.
 - **52** Poulose SM, Miller MG, Shukitt-Hale B. Role of walnuts in maintaining brain health with age. *J Nutr* 2014; 144: 561S–566S.

- **53** Cheong MH, Lee SR, Yoo HS, Jeong JW, Kim GY, Kim WJ, Jung IC, Choi YH. Anti-inflammatory effects of Polygala tenuifolia root through inhibition of NF-kappaB activation in lipopolysaccharide-induced BV2 microglial cells. *J Ethnopharmacol* 2011; 137: 1402–1408.
- 54 Yadav SK, Prakash J, Chouhan S, Westfall S, Verma M, Singh TD, Singh SP. Comparison of the neuroprotective potential of Mucuna pruriens seed extract with estrogen in 1-methyl-4-phenyl-1,2,3,6-tetrahydropyridine (MPTP)-induced PD mice model. *Neurochem Int* 2014; 65: 1–13.
- 55 Ding BJ, Ma WW, He LL, Zhou X, Yuan LH, Yu HL, Feng JF, Xiao R. Soybean isoflavone alleviates beta-amyloid 1-42 induced inflammatory response to improve learning and memory ability by down regulation of Toll-like receptor 4 expression and nuclear factor-kappaB activity in rats. *Int J Dev Neurosci* 2011; 29: 537–542.
- **56** Huang M, Hu YY, Dong XQ, Xu QP, Yu WH, Zhang ZY. The protective role of oxymatrine on neuronal cell apoptosis in the hemorrhagic rat brain. *J Ethnopharmacol* 2012; 143: 228–235.
- **57** Yuan D, Xie YY, Bai X, Wu X, Yang JY, Wu CF. Inhibitory activity of isoflavones of Pueraria flowers on nitric oxide production from lipopolysaccharide-activated primary rat microglia. *J Asian Nat Prod Res* 2009; 11: 471–481.
- **58** Orsi PR, Seito LN, Di Stasi LC. Hymenaea stigonocarpa Mart. ex Hayne: A tropical medicinal plant with intestinal anti-inflammatory activity in TNBS model of intestinal inflammation in rats. *J Ethnopharmacol* 2014; 151: 380–385.
- **59** Mahadevan S, Park Y. Multifaceted therapeutic benefits of Ginkgo biloba L.: chemistry, efficacy, safety, and uses. *J Food Sci* 2008; 73: R14–R19.
- **60** Ramassamy C, Longpre F, Christen Y. Ginkgo biloba extract (EGb 761) in Alzheimer's disease: is there any evidence? *Curr Alzheimer Res* 2007; 4: 253–262.
- 61 Lee KW, Im JY, Woo JM, Grosso H, Kim YS, Cristovao AC, Sonsalla PK, Schuster DS, Jalbut MM, Fernandez JR, Voronkov M, Junn E, Braithwaite SP, Stock JB, Mouradian MM. Neuroprotective and anti-inflammatory properties of a coffee component in the MPTP model of Parkinson's disease. *Neurotherapeutics* 2013; 10: 143–153.
- **62** Olajide OA, Bhatia HS, de Oliveira AC, Wright CW, Fiebich BL. Antineuroinflammatory properties of synthetic cryptolepine in human neuroblastoma cells: possible involvement of NF-kappaB and p38 MAPK inhibition. *Eur J Med Chem* 2013; 63: 333–339.
- **63** Alemany R, Navarro MA, Vogler O, Perona JS, Osada J, Ruiz-Gutierrez V. Olive oils modulate fatty acid content and signaling protein expression in apolipoprotein E knockout mice brain. *Lipids* 2010; 45: 53–61.
- **64** Jung HW, Mahesh R, Lee JG, Lee SH, Kim YS, Park YK. Pinoresinol from the fruits of Forsythia koreana inhibits inflammatory responses in LPS-activated microglia. *Neurosci Lett* 2010; 480: 215–220.
- **65** Dwivedi S, Nagarajan R, Hanif K, Siddiqui HH, Nath C, Shukla R. Standardized extract of bacopa monniera attenuates okadaic acid induced memory dysfunction in rats: effect on Nrf2 pathway. *Evid Based Complement Alternat Med* 2013; 2013: 294501.
- **66** Hsieh PF, Hou CW, Yao PW, Wu SP, Peng YF, Shen ML, Lin CH, Chao YY, Chang MH, Jeng KC. Sesamin ameliorates oxidative stress and mortality in kainic acid-induced status epilepticus by inhibition of MAPK and COX-2 activation. *J Neuroinflammation* 2011; 8: 57.

- 34 Neuroprotective Effects of Phytochemicals in Neurological Disorders
 - **67** Schaffer LF, Peroza LR, Boligon AA, Athayde ML, Alves SH, Fachinetto R, Wagner C. Harpagophytum procumbens prevents oxidative stress and loss of cell viability *in vitro*. *Neurochem Res* 2013; 38: 2256–2267.
 - **68** Jeong K, Shin YC, Park S, Park JS, Kim N, Um JY, Go H, Sun S, Lee S, Park W, Choi Y, Song Y, Kim G, Jeon C, Park J, Lee K, Bang O, Ko SG. Ethanol extract of Scutellaria baicalensis Georgi prevents oxidative damage and neuroinflammation and memorial impairments in artificial senescense mice. *J Biomed Sci* 2011; 18: 14.
 - **69** Ahmed HH, Salem AM, Sabry GM, Husein AA, Kotob SE. Possible therapeutic uses of Salvia triloba and Piper nigrum in Alzheimer's disease-induced rats. *J Med Food* 2013; 16: 437–446.
 - 70 Hwang H, Jeon H, Ock J, Hong SH, Han YM, Kwon BM, Lee WH, Lee MS, Suk K. 2'-hydroxycinnamaldehyde targets low-density lipoprotein receptor-related protein-1 to inhibit lipopolysaccharide-induced microglial activation. *J Neuroimmunol* 2011; 230: 52–64.
 - **71** Lee YJ, Choi DY, Lee YK, Lee YM, Han SB, Kim YH, Kim KH, Nam SY, Lee BJ, Kang JK, Yun YW, Oh KW, Hong JT. 4-O-methylhonokiol prevents memory impairment in the Tg2576 transgenic mice model of Alzheimer's disease via regulation of beta-secretase activity. *J Alzheimers Dis* 2012; 29: 677–690.
 - 72 Lee YJ, Choi DY, Yun YP, Han SB, Kim HM, Lee K, Choi SH, Yang MP, Jeon HS, Jeong JH, Oh KW, Hong JT. Ethanol extract of Magnolia officinalis prevents lipopolysaccharide-induced memory deficiency via its antineuroinflammatory and antiamyloidogenic effects. *Phytother Res* 2013; 27: 438–447.
 - **73** Ulrich-Merzenich G, Kelber O, Koptina A, Freischmidt A, Heilmann J, Muller J, Zeitler H, Seidel MF, Ludwig M, Heinrich EU, Winterhoff H. Novel neurological and immunological targets for salicylate-based phytopharmaceuticals and for the anti-depressant imipramine. *Phytomedicine* 2012; 19: 930–939.
 - 74 Li B, Lee DS, Choi HG, Kim KS, Jeong GS, An RB, Kim YC. Involvement of heme oxygenase-1 induction in the cytoprotective and immunomodulatory activities of Viola patrinii in murine hippocampal and microglia cells. *Evid Based Complement Alternat Med* 2012; 2012: 128019.
 - 75 Ishola IO, Chaturvedi JP, Rai S, Rajasekar N, Adeyemi OO, Shukla R, Narender T. Evaluation of amentoflavone isolated from Cnestis ferruginea Vahl ex DC (Connaraceae) on production of inflammatory mediators in LPS stimulated rat astrocytoma cell line (C6) and THP-1 cells. *J Ethnopharmacol* 2013; 146: 440–448.
 - **76** Khan MM, Kempuraj D, Thangavel R, Zaheer A. Protection of MPTP-induced neuroinflammation and neurodegeneration by Pycnogenol. *Neurochem Int* 2013; 62: 379–388.
 - 77 Jung HW, Kang SY, Park KH, Oh TW, Jung JK, Kim SH, Choi DJ, Park YK. Effect of the semen extract of Thuja orientalis on inflammatory responses in transient focal cerebral ischemia rat model and LPS-stimulated BV-2 microglia. *Am J Chin Med* 2013; 41: 99–117.
 - 78 Terazawa R, Akimoto N, Kato T, Itoh T, Fujita Y, Hamada N, Deguchi T, Iinuma M, Noda M, Nozawa Y, Ito M. A kavalactone derivative inhibits lipopolysaccharidestimulated iNOS induction and NO production through activation of Nrf2 signaling in BV2 microglial cells. *Pharmacol Res* 2013; 71: 34–43.
 - 79 Hosseini M, Mohammadpour T, Karami R, Rajaei Z, Sadeghnia HR, Soukhtanloo M. Effects of the hydro-alcoholic extract of Nigella Sativa on scopolamine-induced spatial memory impairment in rats and its possible mechanism. *Chin J Integr Med* 2015; 21: 438–444.

- **80** Gowran A, Noonan J, Campbell VA. The multiplicity of action of cannabinoids: implications for treating neurodegeneration. *CNS Neurosci Ther* 2011; 17: 637–644.
- **81** Moon M, Choi JG, Kim SY, Oh MS. Bombycis excrementum reduces amyloid-beta oligomer-induced memory impairments, neurodegeneration, and neuroinflammation in mice. *J Alzheimers Dis* 2014; 41: 599–613.
- 82 Zhang S, Qi Y, Xu Y, Han X, Peng J, Liu K, Sun CK. Protective effect of flavonoid-rich extract from Rosa laevigata Michx on cerebral ischemia-reperfusion injury through suppression of apoptosis and inflammation. *Neurochem Int* 2013; 63: 522–532.
- **83** Toldy A, Atalay M, Stadler K, Sasvari M, Jakus J, Jung KJ, Chung HY, Nyakas C, Radak Z. The beneficial effects of nettle supplementation and exercise on brain lesion and memory in rat. *J Nutr Biochem* 2009; 20: 974–981.
- **84** Elango C, Devaraj SN. Immunomodulatory effect of Hawthorn extract in an experimental stroke model. *J Neuroinflammation* 2010; 7: 97.
- 85 Bellezza I, Mierla A, Grottelli S, Marcotullio MC, Messina F, Roscini L, Cardinali G, Curini M, Minelli A. Furanodien-6-one from Commiphora erythraea inhibits the NF-kappaB signalling and attenuates LPS-induced neuroinflammation. *Mol Immunol* 2013; 54: 347–354.
- 86 Garrido-Suarez BB, Garrido G, Delgado R, Bosch F, del CRM. A Mangifera indica L. extract could be used to treat neuropathic pain and implication of mangiferin. *Molecules* 2010; 15: 9035–9045.
- **87** Quartu M, Serra MP, Boi M, Pillolla G, Melis T, Poddighe L, Del Fiacco M, Falconieri D, Carta G, Murru E, Cordeddu L, Piras A, Collu M, Banni S. Effect of acute administration of Pistacia lentiscus L. essential oil on rat cerebral cortex following transient bilateral common carotid artery occlusion. *Lipids Health Dis* 2012; 11: 8.
- 88 Yan Y, Min Y, Min H, Chao C, Ying Q, Zhi H. n-Butanol soluble fraction of the water extract of Chinese toon fruit ameliorated focal brain ischemic insult in rats via inhibition of oxidative stress and inflammation. *J Ethnopharmacol* 2014; 151: 176–182.
- 89 Kao TK, Ou YC, Raung SL, Chen WY, Yen YJ, Lai CY, Chou ST, Chen CJ. Graptopetalum paraguayense E. Walther leaf extracts protect against brain injury in ischemic rats. *Am J Chin Med* 2010; 38: 495–516.
- **90** Tang NY, Liu CH, Hsieh CT, Hsieh CL. The anti-inflammatory effect of paeoniflorin on cerebral infarction induced by ischemia-reperfusion injury in Sprague-Dawley rats. *Am J Chin Med* 2010; 38: 51–64.
- **91** Lee Y, Jung JC, Jang S, Kim J, Ali Z, Khan IA, Oh S. Anti-inflammatory and neuroprotective effects of constituents isolated from Rhodiola rosea. *Evid Based Complement Alternat Med* 2013; 2013: 514049.
- **92** Li H, Liang Y, Chiu K, Yuan Q, Lin B, Chang RC, So KF. Lycium barbarum (wolfberry) reduces secondary degeneration and oxidative stress, and inhibits JNK pathway in retina after partial optic nerve transection. *PLoS One* 2013; 8: e68881.
- **93** Shan Q, Lu J, Zheng Y, Li J, Zhou Z, Hu B, Zhang Z, Fan S, Mao Z, Wang YJ, Ma D. Purple sweet potato color ameliorates cognition deficits and attenuates oxidative damage and inflammation in aging mouse brain induced by D-galactose. *J Biomed Biotechnol* 2009; 2009: 564737.
- **94** Abraham J, Johnson RW. Consuming a diet supplemented with resveratrol reduced infection-related neuroinflammation and deficits in working memory in aged mice. *Rejuvenation Res* 2009; 12: 445–453.

- 36 Neuroprotective Effects of Phytochemicals in Neurological Disorders
 - **95** Singh S, Aggarwal BB. Activation of transcription factor NF-kappa B is suppressed by curcumin (diferuloylmethane) [corrected]. *J Biol Chem* 1995; 270: 24995–25000.
 - **96** Shytle RD, Tan J, Bickford PC, Rezai-Zadeh K, Hou L, Zeng J, Sanberg PR, Sanberg CD, Alberte RS, Fink RC, Roschek B Jr. Optimized turmeric extract reduces beta-Amyloid and phosphorylated Tau protein burden in Alzheimer's transgenic mice. *Curr Alzheimer Res* 2012; 9: 500–506.
 - **97** Park G, Kim HG, Ju MS, Ha SK, Park Y, Kim SY, Oh MS. 6-Shogaol, an active compound of ginger, protects dopaminergic neurons in Parkinson's disease models via anti-neuroinflammation. *Acta Pharmacol Sin* 2013; 34: 1131–1139.
 - **98** Kim S, Lee MS, Lee B, Gwon WG, Joung EJ, Yoon NY, Kim HR. Anti-inflammatory effects of sargachromenol-rich ethanolic extract of Myagropsis myagroides on lipopolysaccharide-stimulated BV-2 cells. *BMC Complement Altern Med* 2014; 14: 231.
 - **99** Bennett L, Kersaitis C, Macaulay SL, Munch G, Niedermayer G, Nigro J, Payne M, Sheean P, Vallotton P, Zabaras D, Bird M. Vitamin D2-enriched button mushroom (Agaricus bisporus) improves memory in both wild type and APPswe/PS1dE9 transgenic mice. *PLoS One* 2013; 8: e76362.
 - **100** Ding H, Zhou M, Zhang RP, Xu SL. [Ganoderma lucidum extract protects dopaminergic neurons through inhibiting the production of inflammatory mediators by activated microglia.] *Sheng Li Xue Bao* 2010; 62: 547–554.
 - 101 Ishii T, Mori T, Tanaka T, Mizuno D, Yamaji R, Kumazawa S, Nakayama T, Akagawa M. Covalent modification of proteins by green tea polyphenol (–)-epigallocatechin-3-gallate through autoxidation. *Free Radic Biol Med* 2008; 45: 1384–1394.
 - 102 Babu PV, Liu D. Green tea catechins and cardiovascular health: an update. *Curr Med Chem* 2008; 15: 1840–1850.
 - **103** Kashima M. Effects of catechins on superoxide and hydroxyl radical. *Chem Pharm Bull (Tokyo)* 1999; 47: 279–283.
 - **104** Nanjo F, Mori M, Goto K, Hara Y. Radical scavenging activity of tea catechins and their related compounds. *Biosci Biotechnol Biochem* 1999; 63: 1621–1623.
 - **105** Lin YL, Lin JK. (–)-epigallocatechin-3-gallate blocks the induction of nitric oxide synthase by down-regulating lipopolysaccharide-induced activity of transcription factor nuclear factor-kappaB. *Mol Pharmacol* 1997; 52: 465–472.
 - 106 Barros D, Amaral OB, Izquierdo I, Geracitano L, do Carmo Bassols Raseira M, Henriques AT, Ramirez MR. Behavioral and genoprotective effects of Vaccinium berries intake in mice. *Pharmacol Biochem Behav* 2006; 84: 229–234.
 - 107 Goyarzu P, Malin DH, Lau FC, Taglialatela G, Moon WD, Jennings R, Moy E, Moy D, Lippold S, Shukitt-Hale B, Joseph JA. Blueberry supplemented diet: effects on object recognition memory and nuclear factor-kappa B levels in aged rats. *Nutr Neurosci* 2004; 7: 75–83.
 - 108 Simonyi A, Wang Q, Miller RL, Yusof M, Shelat PB, Sun AY, Sun GY. Polyphenols in cerebral ischemia: novel targets for neuroprotection. *Mol Neurobiol* 2005; 31: 135–147.
 - **109** Sun AY, Wang Q, Simonyi A, Sun GY. Botanical phenolics and brain health. *Neuromolecular Med* 2008; 10: 259–274.
 - 110 Wang Q, Xu J, Rottinghaus GE, Simonyi A, Lubahn D, Sun GY, Sun AY. Resveratrol protects against global cerebral ischemic injury in gerbils. *Brain Res* 2002; 958: 439–447.

- 111 Solberg NO, Chamberlin R, Vigil JR, Deck LM, Heidrich JE, Brown DC, Brady CI, Vander Jagt TA, Garwood M, Bisoffi M, Severns V, Vander Jagt DL, Sillerud LO. Optical and SPION-enhanced MR imaging shows that trans-stilbene inhibitors of NF-kappaB concomitantly lower Alzheimer's disease plaque formation and microglial activation in AbetaPP/PS-1 transgenic mouse brain. *J Alzheimers Dis* 2014; 40: 191–212.
- 112 Capiralla H, Vingtdeux V, Zhao H, Sankowski R, Al-Abed Y, Davies P, Marambaud P. Resveratrol mitigates lipopolysaccharide- and Abeta-mediated microglial inflammation by inhibiting the TLR4/NF-kappaB/STAT signaling cascade. *J Neurochem* 2012; 120: 461–472.
- **113** Wang HM, Zhao YX, Zhang S, Liu GD, Kang WY, Tang HD, Ding JQ, Chen SD. PPARgamma agonist curcumin reduces the amyloid-beta-stimulated inflammatory responses in primary astrocytes. *J Alzheimers Dis* 2010; 20: 1189–1199.
- 114 Ahlemeyer B, Krieglstein J. Neuroprotective effects of Ginkgo biloba extract. *Cell Mol Life Sci* 2003; 60: 1779–1792.
- **115** Nathan P. Can the cognitive enhancing effects of ginkgo biloba be explained by its pharmacology? *Med Hypotheses* 2000; 55: 491–493.
- 116 Yin Y, Ren Y, Wu W, Wang Y, Cao M, Zhu Z, Wang M, Li W. Protective effects of bilobalide on Abeta(25-35) induced learning and memory impairments in male rats. *Pharmacol Biochem Behav* 2013; 106: 77–84.
- 117 Khan A, Safdar M, Ali Khan MM, Khattak KN, Anderson RA. Cinnamon improves glucose and lipids of people with type 2 diabetes. *Diabetes Care* 2003; 26: 3215–3218.
- **118** Ziegenfuss TN, Hofheins JE, Mendel RW, Landis J, Anderson RA. Effects of a watersoluble cinnamon extract on body composition and features of the metabolic syndrome in pre-diabetic men and women. *J Int Soc Sports Nutr* 2006; 3: 45–53.
- 119 Adisakwattana S, Sompong W, Meeprom A, Ngamukote S, Yibchok-Anun S. Cinnamic acid and its derivatives inhibit fructose-mediated protein glycation. *Int J Mol Sci* 2012; 13: 1778–1789.
- 120 Farr SA, Price TO, Dominguez LJ, Motisi A, Saiano F, Niehoff ML, Morley JE, Banks WA, Ercal N, Barbagallo M. Extra virgin olive oil improves learning and memory in SAMP8 mice. *J Alzheimers Dis* 2012; 28: 81–92.
- 121 St-Laurent-Thibault C, Arseneault M, Longpre F, Ramassamy C. Tyrosol and hydroxytyrosol, two main components of olive oil, protect N2a cells against amyloidbeta-induced toxicity. Involvement of the NF-kappaB signaling. *Curr Alzheimer Res* 2011; 8: 543–551.
- **122** Chung LY. The antioxidant properties of garlic compounds: allyl cysteine, alliin, allicin, and allyl disulfide. *J Med Food* 2006; 9: 205–213.
- 123 Rahman K. Garlic and aging: new insights into an old remedy. *Ageing Res Rev* 2003; 2: 39–56.
- 124 Imai J, Ide N, Nagae S, Moriguchi T, Matsuura H, Itakura Y. Antioxidant and radical scavenging effects of aged garlic extract and its constituents. *Planta Med* 1994; 60: 417–420.
- 125 Lin GH, Lee YJ, Choi DY, Han SB, Jung JK, Hwang BY, Moon DC, Kim Y, Lee MK, Oh KW, Jeong HS, Leem JY, Shin HK, Lee JH, Hong JT. Anti-amyloidogenic effect of thiacremonone through anti-inflamation *in vitro* and *in vivo* models. *J Alzheimers Dis* 2012; 29: 659–676.

- 38 Neuroprotective Effects of Phytochemicals in Neurological Disorders
 - **126** Burdge GC, Calder PC. Conversion of alpha-linolenic acid to longer-chain polyunsaturated fatty acids in human adults. *Reprod Nutr Dev* 2005; 45: 581–597.
 - **127** Cole GM, Frautschy SA. Docosahexaenoic acid protects from amyloid and dendritic pathology in an Alzheimer's disease mouse model. *Nutr Health* 2006; 18: 249–259.
 - 128 Arab L, Liebeskind DS. Tea, flavonoids and stroke in man and mouse. Arch Biochem Biophys 2010; 501: 31–36.
 - 129 Larsson SC. Coffee, tea, and cocoa and risk of stroke. *Stroke* 2014; 45: 309–314.
 - 130 Kokubo Y, Iso H, Saito I, Yamagishi K, Yatsuya H, Ishihara J, Inoue M, Tsugane S. The impact of green tea and coffee consumption on the reduced risk of stroke incidence in Japanese population: the Japan public health center-based study cohort. *Stroke* 2013; 44: 1369–1374.
 - **131** Larsson SC, Virtamo J, Wolk A. Black tea consumption and risk of stroke in women and men. *Ann Epidemiol* 2013; 23: 157–160.
 - **132** Liang W, Lee AH, Binns CW, Huang R, Hu D, Zhou Q. Tea consumption and ischemic stroke risk: a case-control study in southern China. *Stroke* 2009; 40: 2480–2485.
 - **133** Larsson SC, Mannisto S, Virtanen MJ, Kontto J, Albanes D, Virtamo J. Coffee and tea consumption and risk of stroke subtypes in male smokers. *Stroke* 2008; 39: 1681–1687.
 - 134 Tresserra-Rimbau A, Rimm EB, Medina-Remon A, Martinez-Gonzalez MA, de la Torre R, Corella D, Salas-Salvado J, Gomez-Gracia E, Lapetra J, Aros F, Fiol M, Ros E, Serra-Majem L, Pinto X, Saez GT, Basora J, Sorli JV, Martinez JA, Vinyoles E, Ruiz-Gutierrez V, Estruch R, Lamuela-Raventos RM. Inverse association between habitual polyphenol intake and incidence of cardiovascular events in the PREDIMED study. *Nutr Metab Cardiovasc Dis* 2014; 24: 639–647.
 - 135 Mursu J, Voutilainen S, Nurmi T, Tuomainen TP, Kurl S, Salonen JT. Flavonoid intake and the risk of ischaemic stroke and CVD mortality in middle-aged Finnish men: the Kuopio Ischaemic Heart Disease Risk Factor Study. *Br J Nutr* 2008; 100: 890–895.
 - **136** Cassidy A, Rimm EB, O'Reilly EJ, Logroscino G, Kay C, Chiuve SE, Rexrode KM. Dietary flavonoids and risk of stroke in women. *Stroke* 2012; 43: 946–951.
 - 137 Singh N, Agrawal M, Dore S. Neuroprotective properties and mechanisms of resveratrol in *in vitro* and *in vivo* experimental cerebral stroke models. *ACS Chem Neurosci* 2013; 4: 1151–1162.
 - **138** Witte AV, Kerti L, Margulies DS, Floel A. Effects of resveratrol on memory performance, hippocampal functional connectivity, and glucose metabolism in healthy older adults. *J Neurosci* 2014; 34: 7862–7870.
 - 139 Oskouei DS, Rikhtegar R, Hashemilar M, Sadeghi-Bazargani H, Sharifi-Bonab M, Sadeghi-Hokmabadi E, Zarrintan S, Sharifipour E. The effect of Ginkgo biloba on functional outcome of patients with acute ischemic stroke: a double-blind, placebocontrolled, randomized clinical trial. *J Stroke Cerebrovasc Dis* 2013; 22: e557–e563.
 - 140 Zeng X, Liu M, Yang Y, Li Y, Asplund K. Ginkgo biloba for acute ischaemic stroke. *Cochrane Database Syst Rev* 2005: CD003691.
 - 141 Martinez-Gonzalez MA, Dominguez LJ, Delgado-Rodriguez M. Olive oil consumption and risk of CHD and/or stroke: a meta-analysis of case-control, cohort and intervention studies. *Br J Nutr* 2014; 112: 248–259.
 - 142 Samieri C, Feart C, Proust-Lima C, Peuchant E, Tzourio C, Stapf C, Berr C, Barberger-Gateau P. Olive oil consumption, plasma oleic acid, and stroke incidence: the Three-City Study. *Neurology* 2011; 77: 418–425.

- 143 Estruch R, Ros E, Salas-Salvado J, Covas MI, Corella D, Aros F, Gomez-Gracia E, Ruiz-Gutierrez V, Fiol M, Lapetra J, Lamuela-Raventos RM, Serra-Majem L, Pinto X, Basora J, Munoz MA, Sorli JV, Martinez JA, Martinez-Gonzalez MA. Primary prevention of cardiovascular disease with a Mediterranean diet. *N Engl J Med* 2013; 368: 1279–1290.
- 144 Bonds DE, Harrington M, Worrall BB, Bertoni AG, Eaton CB, Hsia J, Robinson J, Clemons TE, Fine LJ, Chew EY. Effect of long-chain omega-3 fatty acids and lutein + zeaxanthin supplements on cardiovascular outcomes: results of the Age-Related Eye Disease Study 2 (AREDS2) randomized clinical trial. *JAMA Intern Med* 2014; 174: 763–771.
- 145 Macchia A, Grancelli H, Varini S, Nul D, Laffaye N, Mariani J, Ferrante D, Badra R, Figal J, Ramos S, Tognoni G, Doval HC. Omega-3 fatty acids for the prevention of recurrent symptomatic atrial fibrillation: results of the FORWARD (Randomized Trial to Assess Efficacy of PUFA for the Maintenance of Sinus Rhythm in Persistent Atrial Fibrillation) trial. *J Am Coll Cardiol* 2013; 61: 463–468.
- **146** Galan P, Kesse-Guyot E, Czernichow S, Briancon S, Blacher J, Hercberg S. Effects of B vitamins and omega 3 fatty acids on cardiovascular diseases: a randomised placebo controlled trial. *BMJ* 2010; 341: c6273.
- 147 Roncaglioni MC, Tombesi M, Avanzini F, Barlera S, Caimi V, Longoni P, Marzona I, Milani V, Silletta MG, Tognoni G, Marchioli R. n-3 fatty acids in patients with multiple cardiovascular risk factors. *N Engl J Med* 2013; 368: 1800–1808.
- **148** Larsson SC, Orsini N, Wolk A. Long-chain omega-3 polyunsaturated fatty acids and risk of stroke: a meta-analysis. *Eur J Epidemiol* 2012; 27: 895–901.
- 149 Poppitt SD, Howe CA, Lithander FE, Silvers KM, Lin RB, Croft J, Ratnasabapathy Y, Gibson RA, Anderson CS. Effects of moderate-dose omega-3 fish oil on cardiovascular risk factors and mood after ischemic stroke: a randomized, controlled trial. *Stroke* 2009; 40: 3485–3492.
- 150 Morillas-Ruiz JM, Rubio-Perez JM, Albaladejo MD, Zafrilla P, Parra S, Vidal-Guevara ML. Effect of an antioxidant drink on homocysteine levels in Alzheimer's patients. J Neurol Sci 2010; 299: 175–178.
- 151 Mastroiacovo D, Kwik-Uribe C, Grassi D, Necozione S, Raffaele A, Pistacchio L, Righetti R, Bocale R, Lechiara MC, Marini C, Ferri C, Desideri G. Cocoa flavanol consumption improves cognitive function, blood pressure control, and metabolic profile in elderly subjects: the Cocoa, Cognition, and Aging (CoCoA) Study – a randomized controlled trial. *Am J Clin Nutr* 2015; 101: 538–548.
- 152 Brickman AM, Khan UA, Provenzano FA, Yeung LK, Suzuki W, Schroeter H, Wall M, Sloan RP, Small SA. Enhancing dentate gyrus function with dietary flavanols improves cognition in older adults. *Nat Neurosci* 2014; 17: 1798–1803.
- **153** Li F, Gong Q, Dong H, Shi J. Resveratrol, a neuroprotective supplement for Alzheimer's disease. *Curr Pharm Des* 2012; 18: 27–33.
- **154** Pasinetti GM. Novel role of red wine-derived polyphenols in the prevention of Alzheimer's disease dementia and brain pathology: experimental approaches and clinical implications. *Planta Med* 2012; 78: 1614–1619.
- **155** Chin D, Huebbe P, Pallauf K, Rimbach G. Neuroprotective properties of curcumin in Alzheimer's disease merits and limitations. *Curr Med Chem* 2013; 20: 3955–3985.
- **156** Hamaguchi T, Ono K, Yamada M. REVIEW: Curcumin and Alzheimer's disease. *CNS Neurosci Ther* 2010; 16: 285–297.

- 40 Neuroprotective Effects of Phytochemicals in Neurological Disorders
 - 157 Ihl R, Tribanek M, Bachinskaya N. Efficacy and tolerability of a once daily formulation of Ginkgo biloba extract EGb 761(R) in Alzheimer's disease and vascular dementia: results from a randomised controlled trial. *Pharmacopsychiatry* 2012; 45: 41–46.
 - 158 Herrschaft H, Nacu A, Likhachev S, Sholomov I, Hoerr R, Schlaefke S. Ginkgo biloba extract EGb 761(R) in dementia with neuropsychiatric features: a randomised, placebo-controlled trial to confirm the efficacy and safety of a daily dose of 240 mg. *J Psychiatr Res* 2012; 46: 716–723.
 - **159** Canevelli M, Adali N, Kelaiditi E, Cantet C, Ousset PJ, Cesari M. Effects of Gingko biloba supplementation in Alzheimer's disease patients receiving cholinesterase inhibitors: data from the ICTUS study. *Phytomedicine* 2014; 21: 888–892.
 - 160 Ihl R, Bachinskaya N, Korczyn AD, Vakhapova V, Tribanek M, Hoerr R, Napryeyenko O. Efficacy and safety of a once-daily formulation of Ginkgo biloba extract EGb 761 in dementia with neuropsychiatric features: a randomized controlled trial. *Int J Geriatr Psychiatry* 2011; 26: 1186–1194.
 - 161 Attia A, Rapp SR, Case LD, D'Agostino R, Lesser G, Naughton M, McMullen K, Rosdhal R, Shaw EG. Phase II study of Ginkgo biloba in irradiated brain tumor patients: effect on cognitive function, quality of life, and mood. *J Neurooncol* 2012; 109: 357–363.
 - 162 Yancheva S, Ihl R, Nikolova G, Panayotov P, Schlaefke S, Hoerr R. Ginkgo biloba extract EGb 761(R), donepezil or both combined in the treatment of Alzheimer's disease with neuropsychiatric features: a randomised, double-blind, exploratory trial. *Aging Ment Health* 2009; 13: 183–190.
 - **163** Mazza M, Capuano A, Bria P, Mazza S. Ginkgo biloba and donepezil: a comparison in the treatment of Alzheimer's dementia in a randomized placebo-controlled double-blind study. *Eur J Neurol* 2006; 13: 981–985.
 - **164** Yang M, Xu DD, Zhang Y, Liu X, Hoeven R, Cho WC. A systematic review on natural medicines for the prevention and treatment of Alzheimer's disease with meta-analyses of intervention effect of ginkgo. *Am J Chin Med* 2014; 42: 505–521.
 - 165 Vellas B, Coley N, Ousset PJ, Berrut G, Dartigues JF, Dubois B, Grandjean H, Pasquier F, Piette F, Robert P, Touchon J, Garnier P, Mathiex-Fortunet H, Andrieu S. Long-term use of standardised Ginkgo biloba extract for the prevention of Alzheimer's disease (GuidAge): a randomised placebo-controlled trial. *Lancet Neurol* 2012; 11: 851–859.
 - 166 Freund Levi Y, Vedin I, Cederholm T, Basun H, Faxen Irving G, Eriksdotter M, Hjorth E, Schultzberg M, Vessby B, Wahlund LO, Salem N Jr., Palmblad J. Transfer of omega-3 fatty acids across the blood-brain barrier after dietary supplementation with a docosahexaenoic acid-rich omega-3 fatty acid preparation in patients with Alzheimer's disease: the OmegAD study. *J Intern Med* 2014; 275: 428–436.
 - 167 Shinto L, Quinn J, Montine T, Dodge HH, Woodward W, Baldauf-Wagner S, Waichunas D, Bumgarner L, Bourdette D, Silbert L, Kaye J. A randomized placebocontrolled pilot trial of omega-3 fatty acids and alpha lipoic acid in Alzheimer's disease. J Alzheimers Dis 2014; 38: 111–120.
 - 168 Faxen-Irving G, Freund-Levi Y, Eriksdotter-Jonhagen M, Basun H, Hjorth E, Palmblad J, Vedin I, Cederholm T, Wahlund LO. Effects on transthyretin in plasma and cerebrospinal fluid by DHA-rich n 3 fatty acid supplementation in patients with Alzheimer's disease: the OmegAD study. *J Alzheimers Dis* 2013; 36: 1–6.

- 169 Vedin I, Cederholm T, Freund-Levi Y, Basun H, Garlind A, Irving GF, Eriksdotter-Jonhagen M, Wahlund LO, Dahlman I, Palmblad J. Effects of DHA-rich n-3 fatty acid supplementation on gene expression in blood mononuclear leukocytes: the OmegAD study. *PLoS One* 2012; 7: e35425.
- 170 Cazzola R, Rondanelli M, Faliva M, Cestaro B. Effects of DHA-phospholipids, melatonin and tryptophan supplementation on erythrocyte membrane physicochemical properties in elderly patients suffering from mild cognitive impairment. *Exp Gerontol* 2012; 47: 974–978.
- 171 Serini S, Bizzarro A, Piccioni E, Fasano E, Rossi C, Lauria A, Cittadini AR, Masullo C, Calviello G. EPA and DHA differentially affect *in vitro* inflammatory cytokine release by peripheral blood mononuclear cells from Alzheimer's patients. *Curr Alzheimer Res* 2012; 9: 913–923.
- 172 Lee LK, Shahar S, Chin AV, Yusoff NA. Docosahexaenoic acid-concentrated fish oil supplementation in subjects with mild cognitive impairment (MCI): a 12-month randomised, double-blind, placebo-controlled trial. *Psychopharmacology (Berl)* 2013; 225: 605–612.
- **173** Yurko-Mauro K, McCarthy D, Rom D, Nelson EB, Ryan AS, Blackwell A, Salem N Jr., Stedman M. Beneficial effects of docosahexaenoic acid on cognition in age-related cognitive decline. *Alzheimers Dement* 2010; 6: 456–464.
- 174 Quinn JF, Raman R, Thomas RG, Yurko-Mauro K, Nelson EB, Van Dyck C, Galvin JE, Emond J, Jack CR Jr., Weiner M, Shinto L, Aisen PS. Docosahexaenoic acid supplementation and cognitive decline in Alzheimer disease: a randomized trial. *Jama* 2010; 304: 1903–1911.
- 175 Freund-Levi Y, Basun H, Cederholm T, Faxen-Irving G, Garlind A, Grut M, Vedin I, Palmblad J, Wahlund LO, Eriksdotter-Jonhagen M. Omega-3 supplementation in mild to moderate Alzheimer's disease: effects on neuropsychiatric symptoms. *Int J Geriatr Psychiatry* 2008; 23: 161–169.
- 176 Barberger-Gateau P, Samieri C, Feart C, Plourde M. Dietary omega 3 polyunsaturated fatty acids and Alzheimer's disease: interaction with apolipoprotein E genotype. *Curr Alzheimer Res* 2011; 8: 479–491.
- 177 Scheltens P, Twisk JW, Blesa R, Scarpini E, von Arnim CA, Bongers A, Harrison J, Swinkels SH, Stam CJ, de Waal H, Wurtman RJ, Wieggers RL, Vellas B, Kamphuis PJ. Efficacy of Souvenaid in mild Alzheimer's disease: results from a randomized, controlled trial. J Alzheimers Dis 2012; 31: 225–236.